AD AO4616	2 1	AD-A046 162 HARVARD UNIV CAMBRIDGE MASS DIV OF ENGINEERING ANDETC F/G 12/2 SCHEDULING DIAL-A-RIDE TRANSPORTATION SYSTEMS: AN ASYMPTOTIC APETC(U) N00014-75-C-0648 NL										
	32							<section-header><section-header><section-header><text></text></section-header></section-header></section-header>				
								 A subscription of the subscriptice of the subscription of the subscription of the subscriptio				
					Antonia de la composición de la composición de la composición de l							
					HARD			A constraint of the second sec			на стала с стала с с с с с с с с с с с с с с с с с с	
												The second secon
						- A CARACTER AND A CA				A second		
					1							

Office of Naval Research Centract H00014-75-C-0648 HR-372-012

National Science Foundation Grant ENG-78-11824

SCHEDULING DIAL-A-RIDE TRANSPORTATION SYSTEMS: AN ASYMPTOTIC APPROACH



DDC CORMAR NOV 2 1977 GGGUU

David M. Stein

September, 1877

Technical Report No. 878

This document has been approved for public release and sale; its distribution is unlimited. Reproduction in whole or in part is permitted by the U.S. Government

Division of Applied Sciences Rervard University Combridge, Nessachusetts

AD NO.

2

AD A 04616

Unclassified SECURITY CLASSIFICATION OF THIS PAGE (When Date Entered) READ INSTRUCTIONS BEFORE COMPLETING FORM REPORT DOCUMENTATION PAGE . REPORT NUMBER 2. GOVT ACCESSION NO. ENT'S CATALOG NUMBER CON Technical Report No.) 679 TITI F (and MOLOVERED SCHEDULING DIAL-A-RIDE TRANSPORTATION Interim Rep **YSTEMS:** AN ASYMPTOTIC APPROACH. . PERFORMING ORG REPORT NUMBER 7. AUTHOR(s) N00014-75-C-06485 R(.) David M. Stein NSF-ENG-76-11824 PERFORMING ONGANY ATTION NAME AND ADDRESS Division of Applied Sciences 10. PROGRAM ELEMENT, PROJECT, TASK AREA & WORK UNIT NUMBERS Harvard University Cambridge, Massachusetts 02138 11. CONTROLLING OFFICE NAME AND ADDRESS . REPORT DATE Sep 1077 148 14. MONITORING ADDRESS(II dillerent from Controlling Office) 15. SECURITY CLASS. (of this report) Unclassified 15. DECLASSIFICATION/DOWNGRADING SCHEDULE This document has been approved for public release and sale; its distribution is unlimited. Reproduction in whole or in part is permitted by the U.S. Government. 17. DISTRIBUTION STATEMENT (of the abstract entered in Block 20, If different from Report) 18. SUPPLEMENTARY NOTES 19. KEY WORDS (Continue on reverse side if necessary and identity by block number) Travelling Salesmen Problem Scheduling Combinatorial Optimization Dial-a-ride Assignment Problem 20. And Act (Continue on reverse elde it necessary and identify by block number) There has been some recent interest in the Dial-a-Ride proposal for innovative public transportation systems. These schemes ideally provide large numbers of passengers with personalized service. A passenger is taken from his origin to his destination by a small bus; the bus may deviate en-route to collect and deliver additional passengers. In such systems the central mathematical problem is one of scheduling: the assignment of a passenger to a bus and the determination of a time for the trip. DD , FORM 1473 EDITION OF I NOV 65 IS OFSOLETE Unclassified 5/N 0102-014-6601 ÷U. TOTALCURITY CLASSIFICATION OF THIS PAGE (Meen Da

Unclassified

CURITY CLASSIFICATION OF THIS PAGE/When Data Entered)

Our investigation is an analytic one into some of the basic aspects of dial-a-ride algorithms. We derive a class of algorithms for which performance can be measured in a certain precise sense. This sense is asymptotic (in the number of passengers who require service) and probabilistic (so that it is likely that on any given day an algorithm will perform well). The approach taken is one of preplanning at a global level. Very loosely, it is based on the following principle. Even though each passenger is unique, with his own required origin, destination and time-of-delivery, in a large system, where there are a large number of passengers, the behavior patterns of the set of all passengers can be predicted quite closely.

To analyse the problem theoretically it is necessary to abstract the essentials and to consider an idealized version. Models of the dial-a-ride problem are presented that focus upon its combinatorial nature. They study separately static version--in which demands for service are all available at the start--and dynamic versions--in which the demands arise over time. For these models, we obtain 'asymptotically optimal' algorithms that minimize simple distance or average flow-time criteria, and we evaluate suboptimal schemes.

Despite the idealization, many qualitative insights ensue. We obtain the implications of the theoretical results for the real problem and propose an approach towards the design of dial-a-ride systems: this approach is based upon the partition of the region into subregions and the specialization of buses to demand-types. The approach is of interest then, because it can be theoretically justified, and also because it has many attractive practical features. For example, resulting systems have modest computational requirements and are simple to visualize and implement. Furthermore, the techniques developed provide an analytical tool for use in the design process. It is possible to investigate changes in performance when parameters (e.g., the number of buses, the size of the region and many others) are varied Proposed schemes can be easily evaluated without the need to resort, basic level, to simulation.

ACCESSION for White Section Buff Section NTIS 200 UNANNOUNCED JUSTI ICA !! DISTRIBUTION AT THE **C**I

Unclassified

Office of Naval Research Contract N00014-75-C-0648 NR-372-012 National Science Foundation Grant ENG 76-11824

SCHEDULING DIAL-A-RIDE TRANSPORTATION SYSTEMS: AN ASYMPTOTIC APPROACH

By

David M. Stein

Technical Report No. 670

This document has been approved for public release and sale; its distribution is unlimited. Reproduction in whole or in part is permitted by the U. S. Government.

September 1977

The research reported in this document was made possible through support extended the Division of Applied Sciences, Harvard University by the U. S. Army Research Office, the U. S. Air Force Office of Scientific Research and the U. S. Office of Naval Research under the Joint Services Electronics Program by Contract N00014-75-C-0648 and by the National Science Foundation under Grant ENG 76-11824.

Division of Applied Sciences

Harvard University · Cambridge, Massachusetts

TABLE OF CONTENTS

D	C	P
-	U.	Ε.

List	of Figures	iv
CHAP	TER I INTRODUCTION	1
1.1	The Dial-a-Ride Problem	2
1.2	Our Approach	8
CHAP	TER II IDEALIZED BUS PROBLEMS: THE STATIC CASE	11
2.0	Idealized Bus Problems	11
2.1	Combinatorial Optimization and the Travelling	15
22	Static Single-Bus Problem	21
	2.2.1 Optimal Algorithms	21
	2.2.2 Comments on the Tour T_m^0	25
	2.2.3 Interesting Suboptimal Algorithms	28
2.3	Static Multiple-Bus Problems	30
	2.3.1 Introduction	30
	2.3.2 Transfers Allowed	33
	2.3.3 No Transfers Allowed	36
CHAP	TER III THE AVERAGE FLOW-TIME CRITERION AND THE	
	DYNAMIC CASE	43
3.1	Static, Single-Bus Problem	43
3.2	The Dynamic Problem; Single-Bus Case	49
3.3	Dynamic, Multiple-Bus Case; Transfers Allowed	53
3.4	Dynamic, Multiple-Bus Case; Transfer-Free	58
	3.4.1 Two-Bus Schemes	59
	J. J. Z A INFEE-BUS Scheme	01
CHAP	TER IV TOWARDS THE PRACTICAL DESIGN OF	
	DIAL-A-RIDE SYSTEMS	65
4.1	Introduction	65
4.2	A Design Approach	66
4.3	Quantitative Aspects of Design	79
	4.3.1 The Demand Distribution	91
	4.3.2 Acumptotic Approximation	82
	4.3.4 Additional Uncertainty	89
4.4	An Example	91
CHAP	TER V CONCLUSION	101
APPE	NDIX TO CHAPTER II	105
APPE	NDIX TO CHAPTER III	127
REFE	RENCES	139
1		

LIST OF FIGURES

PAGE 1.1 The 'Michigan Scheme' 5 2.1 Partition of R into subregions 29 33 2.2 Algorithm 3 37 2.3 Two-bus static scheme without transfers 2.4 Three-bus static scheme without transfers 39 59 3.1 Two-bus dynamic scheme without transfers 3.2 Three-bus dynamic scheme without transfers 62 4.1 Random instances of the travelling salesman 85 problem, generated on a unit square 4.2 Asymptotic convergence to Beardwood's formula 86 4.3 The rate of convergence of $E[L_n]/\sqrt{n}$ 87 4.4 An example: the city 92 4.5 Partition of suburbs, Rs 93 4.6 Route taken by a single bus in a region 94 97 4.7 Route taken by a pair of buses in a region 98 4.8 Decomposition for suburban demands A2.1 Diagram for Lemma 2.1 A2.2 The tour T_n^{\bullet} - (i) 109 115 A2.3 The tour T_n^* - (11) 117 A2.4 Representation of tours and transfers by a 123 graph

.

CHAPTER I

INTRODUCTION

This report is, in essence, concerned with scheduling theory. The concern takes two forms. First, there is a practical engineering scheduling problem that needs to be solved. To study this problem we have drawn upon a new theoretical approach to scheduling. And, this is our second concern, the development of a theory that indeed has practical implications.

The problem we address is that of scheduling dial-a-ride transportation systems. A solution would be the development of a methodology for analysing and designing the scheduling algorithms. Whilst aiming at a solution, we investigate to what extent a recent asymptotic probabilistic technique for the solving of hard combinatorial optimization problems is of real interest. This requires that a mathematical result be generalized for a number of problem formulations.

-1-

1.1 The Dial-a-Ride Problem

During the past decade there has been some interest in the planning of innovative public transportation systems. An area which has received substantial attention is that of 'demand responsive transportation' [19].

One of the outcomes of this research has been the 'dial-a-ride' proposal. A dial-a-ride transportation system is somewhere in the range between a rigid bus system and a flexible taxicab system, and ideally provides large numbers of passengers with personalized service. Passengers request service - to be taken from an origin to a destination - by telephone. At this time of request, an estimate of the time of collection and time of delivery is quoted. Small buses travel about the region, collecting and delivering passengers; during his trip, a particular passenger may be transferred to another bus. At any time a bus may have many passengers aboard.

The central mathematical problem in a dial-a-ride transportation system is one of scheduling: an algorithm is required that will decide to which bus (or sequence of buses) a particular passenger should be assigned and when his trip should take place. A number of algorithms have been proposed to do this scheduling. Most of these are computer aided; the computer is used for the detailed on-line decision making or as an information source, evaluating the system 'state'.

-2-

We shall be concerned with dial-a-ride scheduling algorithms at a fundamental level, avoiding many of the modifications and adjustments that must, of course, be made in an actual implementation (we shall discuss some of the practicalities in Chapter IV). There are two fundamentally very different approaches to scheduling which we shall consider. Let us describe the skeletons of these approaches.

The first is a system developed at M.I.T. by Wilson et. al. [21,22]; experimental versions have been mented since 1972, and currently one is being tested at chester, New York. The underlying trait of this scheme is its search procedure for allocating passengers to buses. At each instant of time each bus has associated with it a 'prospective route', given by an ordered sequence of future stops (which may be origins or destinations (1)), and an estimated time-of-arrival at each of these stops. With each stop too there is a 'latest time-of-arrival' - the time which has been quoted to the associated passenger. A new incoming passenger must be allocated to a bus. This involves inserting his origin and destination into the prospective route of one of the buses. For each new demand, the scheduler searches through every possible insertion on every one of the buses and chooses the best, so as to

(1) This scheme requires no transfers for passengers: each passenger must be delivered by the same bus that collects him.

-3-

minimize a certain criterion function⁽²⁾. Basic to the design of algorithms here is the choice of the criterion function, the method of quotation, and the travel-time prediction. It appears that the procedure is very sensitive to these choices, as well as to variations in the parameters [21, 22, 18].

At the other extreme is a very different approach towards the scheduling. It is described by the following, which we shall refer to as the 'Michigan Scheme' since a related system has been evolving at Ann Arbor, Michigan [17]. The region in which the system operates is partitioned into a number of subregions, $r_1, r_2, \ldots r_m$. In each region r_1 there is a bus that travels only in r_1 . A

(2) The criterion is a function that concerns both present and future passengers. Present passengers are interested in their wait time, w, ride-time, r, total travel time, w+r, pickup-time deviation, Dp, and delivery-time deviation, Dd (these are deviations from the quoted times). Future passengers are acknowledged by taking into consideration the increase in tour-length, DT. Thus, for each possible assignment one can evaluate

 $a_1f_1(w) + a_2f_2(r) + a_3f_3(w+r) + a_4f_4(Dp) + a_5f_5(Dd)$ (1)

for every current passenger, and a similar value for the passenger being assigned. Here, the ai's are weighting parameters and the f_i 's are functions (generally, $f_i(x) = x$ or x^2 for all i). Summing (1) for all the passengers and adding also bDT

where b is another weight, yields the criterion function to be minimized.

The time to a point P which is quoted a passenger is determined by $c.E(t_p) + d$

where $E(t_p)$ is the expected direct travel-time to P and c and d are fixed parameters. In some versions of the algorithm the quoted times are retained as hard constraints, which must never be violated by subsequent assignments.

-4-

larger 'line-haul' bus connects the regions by travelling along a fixed route, stopping at certain 'transfer points', $p_1, p_2, \ldots p_m$ in each region. This is shown schematically in Figure 1.1.



Figure 1.1 The "Michigan Scheme"

A passenger requiring to go from region r_i to region r_j is collected by the bus in r_i , transfers onto the line-haul bus at p_i , alights at p_j and is delivered by the bus in r_j . The regional buses visit their transfer points every 15 minutes (say). The regions are small enough that drivers can quite easily choose an optimal or near-optimal path between visits to their transfer point. The times which are quoted to passengers for collection and delivery are simple estimates of travel time given the load⁽³⁾. Under a heavily loaded system a passenger may have to wait his turn before being collected, as the regional

(3) These quotations are not used in the scheduling procedure.

- 5-

buses can make only a limited number of stops between visits to the transfer point.

We do not intend to give here a detailed evaluation of present dial-a-ride algorithms - the reader is referred to [10] - but some critical comments are in order. These focus on three aspects: the cost, the optimality and the practical performance. We base our initial observations on the Rochester approach, since there has been a great deal of research into this and related schemes.

Basic to the Rochester approach is a long and hard search. Typically, the computational effort required for such searches grows exponentially with the problem size (see the comments in Section 2.1), and we should expect it to be very expensive.

A scheme might be justified if a certain performance level can be guaranteed. The Rochester search is a local one [5], both with respect to time and with respect to the passengers. That is, only a single passenger is assigned (or at best only a small set of passengers is reassigned) when the schedule is updated. Further, optimality (with respect to the criterion function) is ensured for only that point in time. For local searches it is very rare that good performance can be guaranteed. Recent research in [22] has focused on the criterion function. However, there is no guarantee that the performance of the system as a whole (measured, say, in terms of average travel-time and/or

-6-

variance of travel-time⁽⁴⁾) will be improved if the utility functions of all current passengers are maximized whenever decisions are made. This is true even if reasonably acceptable utility functions for the passengers could be represented, itself a notoriously messy problem.

Lastly, what about its practical performance? In the early stages of implementation, particularly when the system was heavily loaded, there were often roundabout devious routes for passengers, with resulting customer dissatisfaction [18]. In general, it appears that the results on performance are inconclusive.

In contrast, the simplicity of the scheme at Michigan is very attractive. The scheduling cost is relatively low, since the computer is used only to store the demands and access them efficiently. (Optimal tours for the regional buses might also be determined on-line by a computer. Then each tour is but a small, simple 'travelling salesman tour' - see Chapter II - where the distances between points satisfy the triangle inequality. For these problems there are some recent efficient heuristic algorithms [12].)

In practice the Michigan scheme has been found to work well [17]. But there are still some vague questions. Can one justify the use of the Michigan approach with respect to some optimality criterion; and how should one design the regions, the fixed routes and the timing of transfers?

(4) These were the measures loosely used in [22].

-7-

1.2 Our Approach

Our investigation will be an analytic one into some of the basic aspects of dial-a-ride algorithms. We shall derive a class of algorithms - or, more correctly, an approach towards the design of algorithms - for which optimality can be measured in a certain precise sense. This sense is asymptotic (in the number of passengers who require service) and probabilistic (so that the probability is high that on any given day the algorithm will perform well).

The approach taken is one of preplanning at a global level. Very loosely, it is based on the following principle. Even though each passenger is unique, with his own required origin, destination and time-of-delivery, in a large system, where there are a large number of passengers, we can predict quite closely the behaviour patterns of the set of all passengers. This is the 'equalizing' effect of the law of large numbers that has been observed in many physical phenomena, the classic example being the thermodynamic principles for the behaviour of gases.

To analyse the problem theoretically it is necessary to abstract the essentials and to consider an idealized version. Chapters II and III present an analysis of models of the dial-a-ride problem that focus upon its combinatorial nature. They study separately static versions - in which demands for service are all available at the start - and dynamic versions - in which the demands arise over time.

-8-

For the models, we obtain 'asymptotically optimal' algorithms that minimize simple distance or average flow-time criteria, and we evaluate suboptimal schemes.

Despite the idealization, many qualitative insights result, and in Chapter IV, we return to the real problem. Here, drawing upon the theoretical results, we obtain their implications for the real problem and propose an approach towards the design of dial-a-ride systems. The approach is of interest then, because it can be theoretically justified, and also because it has many attractive practical features. Furthermore, the techniques developed in Chapters II and III provide us with a powerful analytical tool for use in the design process. We can investigate changes in performance when parameters (e.g. the number of buses, the size of the region and many others) are varied. Proposed schemes can be easily evaluated without the need to resort, at this basic level, to simulation.

It is interesting to note, at this point, that the 'Michigan Scheme' described above can be considered as belonging to our class of algorithms. Thus, we are able to say in what sense it is optimal as well as to gain some valuable insights towards its improvement. The attractive simplicity of the Michigan scheme is common to all of our proposed algorithms.

-9-

CHAPTER II

STATIC BUS PROBLEMS

In this chapter we begin a theoretical analysis of dial-a-ride bus systems. In Section 2.0, we introduce the idealized 'bus problems' which we shall be developing, and state the basic assumptions and terminology. Our approach is based on an interesting theorem of Beardwood, Halton and Hammersley [2] and on a recent paper of Karp [9]; we describe these results within the context of combinatorial optimization problems in Section 2.1. Sections 2.2 and 2.3 present a theory for single- and multiple-bus problems.

Throughout this chapter the presentation is informal, with only heuristic justifications and proofs. Detailed proofs of the results are delegated to an appendix.

2.0 Idealized Bus Problems

The travelling salesman problem (TSP) - see the definition below - has received much attention in the operations-research literature. Not least among the reasons for this is the fact that the TSP can be regarded as a prototype for many realistic problems. Clearly, there is a nontrivial relationship between the TSP and the dial-a-ride

B PAGE BLANK-NOT FILMED

problem, and in this chapter we shall exploit this relationship. First let us describe the problems to be considered, and state the basic assumptions and our terminology.

We are given a bounded region R in the plane, with area a. Demands arise in R. A demand p = (o,d) is a pair of points in R with origin o and destination d. A set of m buses is at our disposal for meeting these demands - i.e. for visiting the set of points. A tour for a bus is the sequence of points which it visits. Before a bus can visit a destination the relevant passenger must be on board, and so the corresponding origin must already have been visited. We refer to this as the 'feasibility constraint'; tours which satisfy it are termed 'feasible'.

We distinguish between static and dynamic versions of the problem. In the static version we are given, at time t=0, a collection of n demands and we are to devise feasible tours for the buses so that all 2n points are visited. In dynamic problems, to be discussed in Chapter III, the demands arise as time progresses according to some random process. Again feasible tours are to be devised (of course a point can be visited only after the demand arises). The problem faced is to devise tours so that a certain criterion is minimized. The criterion will usually be a function of time or of distance travelled.

-12-

We assume that the buses are all of infinite size. The distance between two points in R is the euclidean length of the straight line joining them, so if R is not convex the tour might leave R. The buses travel at unit speed, and no time is wasted when a passenger embarks or alights. Also, transfers are instantaneous (see Section 2.3). These assumptions will be discussed further in Chapter IV.

An instance of a static bus problem of size n is specified by a set of n demand pairs. Our approach is probabilistic and we must define a probability distribution from which the problem instances are drawn. We assume (for simplicity, although this can be considerably weakened) that all origins and destinations are drawn independently from the uniform probability distribution over R. We refer loosely to this fact by saying that a particular problem instance (of size n) is 'random'.

We shall also be making use of the well-known travelling salesman problem. An instance of this problem (the euclidean version) is given by a set of n points within R. We are required to find a path⁽¹⁾ which passes through these n points and which has shortest length. In a 'random' instance of the TSP, the n points are drawn independently from a uniform probability distribution over R.

(1) By a 'travelling salesman problem' we mean this 'open' version, in which the required tour through the n points need not be closed.

-13-

In the algorithms to be considered, the region R is partitioned into subregions (this emphasis will be justified). When we say a bus 'visits' certain points in a subregion we mean that it enters the subregion and performs an optimal travelling salesman tour on the designated points there. When the bus, then visits points in another region, it travels to the closest of the new points and similarly performs an optimal travelling salesman tour in that region.

If we think of R as being divided into m equal subregions each of area a/m, then any given demand pair originates in any particular region with probability 1/m, and has destination in any region with probability 1/m. If there are n demands, with n a large number, then with high probability there are approximately n/m origins and n/m destinations in each region. This follows from the strong law of large numbers [14]. However, we might loosely assume that there are exactly n/m origins and n/m destinations in each region. This is an example of the sort of imprecision that exists in the following pages. Rigorous statements and detailed proofs of the results are collected in the appendix, so we are free to concentrate here on the spirit of the ideas. It should be kept in mind throughout that inexactitudes as that above hold only asymptotically (in n) with probability 1.

-14-

Our analysis will begin with a very simple bus problem, and the results will then be extended to more complex problems with corresponding increased realism. First, however, we digress to describe the aspects of scheduling theory upon which our approach is based.

2.1 Combinatorial Optimization and the Travelling Salesman Problem

Recently there have been some theoretical advances in the understanding of combinatorial optimization problems. Most important has been the acknowledgement that there exists a class of 'hard problems' (NP-complete problems). These are hard in the sense that the computing time required to find their solution by any known algorithm explodes exponentially as the size of the problem increases. For these hard problems, then, there seems to be no way to avoid, essentially, the enumeration of a very large number of possible alternatives. This is inefficient; an 'efficient' algorithm would be one with an execution time behaving as p(n), a polynomial function of n. These notions have been made mathematically precise - see Aho et. al. [1], Coffman [5].

For example, the TSP has been shown to be NP-complete [16], and there are no known efficient algorithms for its solution. To solve it, we need to evaluate very many of the n/21 possible orderings of the points, or tours. Furthermore, the existence of an efficient algorithm for the TSP would imply, and be implied by, the existence of efficient algorithms for a whole class of hard scheduling and other combinatorial optimization problems. At present, this appears extremely unlikely.

This is indeed bad news, but practical problems must still be solved. To this end, note that we required above that the algorithm guarantee the optimal solution. An easier question which might be asked would be one with the optimality requirement relaxed. So, find an efficient algorithm A (if one exists!) that will provide a possibly nonoptimal cost - call it c(A) - but one that is close to the optimal cost, c^* . More precisely, an algorithm A is said to solve the problem to within the ratio y (y is a real number, larger than 1) if

 $c(A) \leq yc^{*}$ (2.1) (We shall also use the terminology: A is ε -optimal if $c(A) \leq (1 + \varepsilon)c^{*}$)

This approach has been found to be very useful for certain special problems, and of no use at all for certain others. For the travelling salesman problem the best that has been achieved in this sense is an algorithm (Christofides [3]) that solves the problem in polynomial time to within 3/2. This ratio is still a little high and one mignt expect to do better. For, given an instance of the travelling salesman problem, it is not too hard to draw a path which looks pretty much the shortest, and one might then be tempted to say that with high probability it is close to the optimum. This implies that we should relax not only the optimality requirement but also the guarantee which the algorithm promises. This notion coincides with a very recent approach, suggested by Karp [9]. Suppose that the problem instance is derived from a certain known probability distribution. Can we then produce an efficient algorithm that will perform well (i.e. to within an acceptable ratio) with high probability? This probabilistic approach can best be illustrated via the TSP.

Let us suppose that a particular problem instance of the TSP is chosen by drawing n points independently from a uniform probability distribution over the planar region R. Let L_n be the length of the shortest path through these n points. Then L_n is a random variable.

The following theorem⁽²⁾ is due to Beardwood et. al. [2].

(2) The problem studied in [2] is actually the <u>closed</u> TSP in which a closed tour is sought. It is easy to see that the theorem still holds for the open TSP.

-17-

Theorem 1

If the region R is bounded and has area a, then there exists an absolute constant b such that

 $\lim_{n \to \infty} \frac{L_n}{\sqrt{n}} = b\sqrt{a} \quad \text{almost everywhere (a.e.) (2.2)}$

The existence of the limit in (2.2) is part of the assertion. The constant b has been estimated by Monte Carlo experiments to be .75. Of course, the rate of convergence of the sequence L_n/\sqrt{n} is important; this aspect of the problem will be discussed in detail in Section 4.3. We shall also discuss there the case in which the probability distribution is not uniform.

By Theorem 1, then, the length of the shortest path, a random variable, is asymptotically (i.e. for n large) equal to $b\sqrt{a}\sqrt{n}$ with probability 1. The value $b\sqrt{a}\sqrt{n}$ is a non-random function of n and so for large n there is no distinction between the random and non-random versions of the problem, and we can predict with probability 1 the length of the optimal tour through any random points. In particular, suppose that an algorithm A yields a tour length of L_n^A through n random points, and that n is very large. Then, by definition (2.1), this algoritm solves the problem to within $L_n^A/b\sqrt{a}\sqrt{n}$ with probability 1. It is in this asymptotic probabilistic sense that we are able to investigate the optimality of various algorithms.

-18-

Consider the following algorithm (similar to that of Karp [9]) which yields a path for any instance of the TSP.

Algorithm 1

Divide the region R into m subregions each of area a/m, and label them $r_1, r_2, \ldots r_m$. Using an optimal algorithm, construct an optimal travelling salesman tour within each of these regions individually. Now, visit the regions $r_1, r_2, \ldots r_m$ in order; upon completing a tour in region r_i , visit the closest unvisited point in region r_{i+1} (take $r_{m+1} = r_1$) and thence traverse the tour in that region.

For n large enough there will be n/m points in each region; the tour within each region has length⁽³⁾ (by Theorem 1)

 $b \prod_{m} a = b \sqrt{n} \sqrt{a}$ (a.e.).

Calling the total distance T_n^K we have

 $T_n^K \leq m b \sqrt{n} \sqrt{a} + m \Delta$ (a.e.)

(3) It should be clear that these equations and inequalities hold in the asymptotic probabilistic sense only. The derivation can easily be made rigorous. The appendix of Chapter II will illustrate how this can be done. where Δ is the diameter⁽⁴⁾ of the region R. Hence,

$$\frac{\underline{T}_{\underline{n}}^{\mathbf{A}}}{\sqrt{n}} \leq b\sqrt{a} + \underline{m}\Delta \qquad (a.e.)$$

and, given any ε , for n large enough,

$$\frac{T_n^K}{\sqrt{n}} \leq b\sqrt{a} + \varepsilon \qquad (a.e.)$$

So, loosely speaking, Algorithm 1 is 'asymptotically optimal'. More formally, we have

Corollary 1

Given any $\varepsilon > 0$, there exists an N(ε) such that for any random problem of size n, with $n \ge N(\varepsilon)$, Algorithm 1 is ε -optimal with probability 1. []

It can further be easily shown - see Karp [9] - that if **m** grows as loglogn (so that $m/n \rightarrow 0$ as $n \rightarrow \infty$) and an $O(n.2^n)$ algorithm (e.g. that in [8]) is used within each subregion, then Algorithm 1 runs in time O(nlogn) a.e.

(4) The diameter of R is $\min\{||x-y||; x, y \in R\}$. The term involving the diameter becomes negligible in the limit. The formulas are independent of the shape (or convexity) of the region. For example, in a region with shape

the travelling salesman tour will still be hardly likely to exit from the region if n is large enough.

-20-

2.2 Static, Single-Bus Problems

2.2.1 Optimal Algorithms

The simplest static bus problem is one in which there is only a single bus. We wish to choose a feasible tour through the n demand pairs so as to minimize the total distance travelled (this will be the same as minimizing the time by which the final passenger is delivered). This problem differs from the TSP only in the feasibility constraint.

As a first observation, note that the problem is NP-complete (see Observation 2.0 in Appendix II).

Let us define (5)

 L_n = length of the optimal travelling salesman tour through n random points in R.

 Y_n = length of the optimal reasible bus tour through n random demand pairs in R.

Observe that, by dropping the feasibility constraint, we would obtain a tour of length L_{2n} . Hence

$$L_{2n} \leq Y_n$$
 (2.3)

An upper bound to Y_n can be obtained by any suboptimal algorithm which might be suggested. For example, if we first visit all origins in R and then visit all destinations, we would obtain a tour with length $L_n^1+L_n^2$,

(5) Both L_n and Y_n are random variables.

-21-

where L_n^1 and L_n^2 are the lengths of the two travelling salesman tours on n random points each. This yields

 $Y_n \leq 2L_n$.

We can easily improve on this upper bound as follows. Divide R into two equal subregions of area a/2 each, calling them regions r_1 and r_2 . First, visit all origins in r_1 (there will be n/2 points, from demands of the form (r_1,r_1) and (r_1,r_2) for n large). Second, visit the origins and the destinations from demands (r_1,r_2) in r_2 (there will be 3n/4 points here). Third, visit r_1 again, visiting the remaining points there - these will be the n/2 destinations from demands (r_1,r_1) and (r_2,r_1) . Finally, visit the last destinations in r_2 - there will be n/4 points from demands (r_2,r_2) . For n large this tour has approximate length (a.e.)

> $b(\sqrt{n/2} + \sqrt{3n/4} + \sqrt{n/2} + \sqrt{n/4})\sqrt{a/2}$ = 1.96b $\sqrt{a}\sqrt{n}$ = 1.96L_n.

The obvious generalization is easy, and we can give the following algorithm, producing a feasible tour which we call T_m^0 .

Algorithm 2 (producing the tour T_m^0)

Partition the region R into m equal subregions, each of area a/m, and label them $r_1, r_2, \ldots r_m$. Visit the regions in numerical order; in each region r_1 collect all the origins as well as the destinations from regions $r_1, r_2, \ldots r_{i-1}$. This is the 'first passage' through the

-22-

regions. Following this, again visit the regions $r_1, r_2, \ldots r_m$ in order, visiting all the remaining destinations in each. This is the 'second passage' through the regions. Each time a region is visited, the tour is a travelling salesman tour. []

It is shown in Lemma 2.2 that the tour T_m^0 has length $\frac{4}{3}\sqrt{2}b\sqrt{a}\sqrt{n} + \sqrt{n}O(1/m)$ (2.4)

If we now assume, as appears reasonable at this stage, that there exists an absolute constant c such that

 $\lim_{n \to \infty} \frac{Y_n}{\sqrt{n}} = c\sqrt{a} \qquad (a.e.) \qquad (2.5)$

then (2.4) and (2.5) (noting that $L_{2n} = \sqrt{2n}b\sqrt{a} = \sqrt{2}L_n$) together give

JZ ≤ c ≤ 4JZb/3.

More difficult is Lemma 2.3 where it is shown that the value of $4\sqrt{2}b/3$ is also a lower bound to c (the rigorous proofs in the appendix do also guarantee the existence of the limit in (2.5) and hence too that c exists). We give a rough outline to the proof of Lemma 2.3. Given any optimal tour with length Y_n , it is possible to construct a suitable division of the area R into m equal subregions (for any m \geq 1) such that the following infeasible tour does yield a lower bound to the value Y_n . This infeasible tour visits the regions successively, twice each, in exactly the manner of the tour T_n° , except that the destinations of the

form (r_1, r_1) , i=1, 2, ..., m, are visited in each region r_1 on the first passage through the regions and not on the second⁽⁶⁾. Letting $m \rightarrow \infty$ the lengths of the infeasible tours converge to $4\sqrt{2}b\sqrt{n}/3$, a.e.

From Lemmas 2.2 and 2.3 we can state

Theorem 2

 $\lim_{n \to \infty} \frac{Y_n}{\sqrt{n}} = \frac{4}{\sqrt{2}} \sqrt{2} \sqrt{a} \qquad \text{a.e.}$ $= 1.89 \sqrt{a}.$

[]

Henceforth for convenience we write

c = 4/2b/3.

We shall also refer to the 'optimal tour T° ' with length $c\sqrt{n}\sqrt{a}$, where $T^{\circ} = \lim_{m \to \infty} T^{\circ}_{m}$; by this rough statement we mean that we can approach arbitrarily close to the optimal value of Y_n by a tour T°_m given by Algorithm 2; this is still asymptotic in n (with $n/m \to \infty$) a.e.

More precisely, we get from Lemma 1,

(6) Note that if there are n_i points to be visited in a certain region r_i and if n'_i and n''_i are visited on the first and second passages through r_i respectively, then the tour in r_i has length behaving as

 $y_1 = b\sqrt{a/m}(\sqrt{n_1^1} + \sqrt{n_1^n})$

with $n_1^* + n_1^* = n_1$

The value y_1 is minimized if $|n_1'-n_1'|$ is made as large as possible.

Corollary 2(7)

Given any $\varepsilon > 0$, there exist $N(\varepsilon)$ and $m(\varepsilon)$ such that for $n \ge N$, the tour T_m^0 given by Algorithm 2 minimizes the total distance travelled to within $1+\varepsilon$, with probability 1.

2.2.2 Comments on the Tour Im

(i) Uniqueness of the Tour T^o

Note that the tour T° described above does not uniquely solve the minimum distance problem. Consider a tour T_m° which visits the regions $(r_1, r_2, \ldots r_m)$ in the same manner as T_m° on the first passage through the regions. On the second passage, however, the regions are visited in the order $(r_m, r_{m-1}, \ldots r_1)$, and once again all feasible destinations are visited in each region. Clearly T_m° and T_m° have the same length (asymptotically a.e.) and

$$T^{\circ'} \stackrel{\Delta}{=} \lim_{m \to \infty} T_m^{\circ'}$$

would also solve the problem.

(7) This statement can be strengthened slightly: Given any $\varepsilon > 0$ there exists an $M(\varepsilon)$ and a function $N(m, \varepsilon)$ such that for all $m \ge M$, the tour T_m^0 minimizes the distance to within $(1+\varepsilon)$ a.e., whenever $n \ge N(m, \varepsilon)$.

In later sections, particularly for the dynamic problem, we shall be interested in other criteria; there is then an inefficiency in the tour $T_m^{O'}$ as it is 'tail-heavy' with passengers having on the average a longer travel-time. This justifies our present emphasis on the tour T_m^{O} .

There is yet another tour which minimizes the distance travelled. Consider visiting regions $(r_1, \ldots r_m)$ in the same order on two separate passages. On the first, collect in each region r_i all origins of the form (r_i, r_j) , $j \leq i$. On the second passage, collect all origins of the form (r_i, r_j) , j > i and simultaneously deliver all feasible destinations (these will be all of the destinations in r_i). It is easy to see that this tour will have the same length as the tour T_m^o , but will be even more tail-heavy than was $T_m^{o'}$.

We might emphasise the easy principle underlying the optimal tour T_m^0 , which will later be seen to be important. Visit the regions successively; each time the bus exits from a region choose as next region the one with the most unvisited feasible points (break ties arbitrarily), and visit all of these points.

(11) Simple Tours

This is a suitable juncture at which to digress to make a further important observation on the tour T° . It belongs to a class of tours which we shall term 'simple'.

-26-

Conceptually, a simple tour is one that visits the points in R region-by-region. The tour segment in r_i can be constructed at the time at which the bus enters r_i , in that all points visited in r_i were feasible at this time of entry.

More precisely, let m be any integer, $m \ge 1$. Define S_m , the set of 'simple tours on m subregions' as follows. A feasible tour T belongs to S_m if there exists a partition of R into at least m subregions of area a/m each, with the two properties:

1) if T enters a region r_1 at time t_1 , then the points which are visited in r_1 were already feasible at time t_1 , 2) if T visits p points in r_1 , then we can assume that these p points are randomly distributed over r_1 (and that T performs an optimal travelling salesman path on these p points).

Now define

$$S = U S_{i}$$

Then any tour in S is feasible, and

 $S_i \subseteq S_{i+1} \subseteq S$ for any i=1,2,...We call S the set of 'simple tours'.

By Corollary 2, for the single-bus static problem it is enough to consider only simple tours. Most tours which we encounter will be simple; in Chapter III we shall restrict our consideration to simple tours only.

-27-

2.2.3 Interesting Suboptimal Algorithms

It is of interest to describe briefly some suboptimal algorithms. These are closely related to currently used techniques and will be needed subsequently.

The following might be considered to be the 'Michigan Algorithm', described in Chapter I, for the static single-bus case.

Partition the region R into m subregions. Calculate an optimal travelling salesman tour on the origins within each region. Visit all regions, performing this optimal tour in each, and linking the regions together using some fixed-route, of any length g. Then, again return along the same fixed-route, deviating within each region to deliver all destinations, using an optimal travelling salesman tour within each region.

Let Y_n^M be the total length of this bus tour. Then

$$n_n = 2mL_{n/m} + 2g$$

and

 $\lim_{n \to \infty} \frac{Y_n^M}{\sqrt{n}} = 2b\sqrt{a} \qquad a.e.$

The distance g travelled by the fixed-route bus is asymptotically negligible. It will henceforth be convenient to think of the region R as the interior of a circle, and the subregions as sectors (see Figure 2.1).

-28-



Figure 2.1 Partition of R into subregions

It is then easy to visualize, for example, a bus visiting the subregions in any required order, and to eliminate the need for a fixed-route bus. However, this convenience is not necessary.

The 'Michigan tour' described above is now seen to be only a very simple extension of Algorithm 1 to the bus problem. With respect to distance travelled, it can be considered the same as the tour T_1^0 given by Algorithm 2.

Most important, from the practical point of view, is that (see (2.1)) the algorithm solves the problem to within 2b/c = 1.06 a.e.

A second suboptimal algorithm is the following 'fixed-route' algorithm. Again the region is divided into m subregions and a priori it is specified that the regions will be visited first in the order $r_1, \ldots r_m$ and then in the order $r_m, r_{m-1}, \ldots r_1$. On the first passage, origins of the form (r_i, r_j) , $i \ge j$ are collected and destinations of the form (r_i, r_j) , i < j are delivered. In region r_i , there will be $(m-i+1)n/m^2$ origins and $(i-1)n/m^2$ destinations visited, a total of n/m points. On the second passage all remaining origins and destinations are visited in each region (this will be feasible).

Let Y_n^F be the total length of this bus tour. Then, $Y_n^F = 2mb \int_m \int_m$

or, more precisely,

 $\lim_{n \to \infty} \frac{\underline{Y}_n^F}{\sqrt{n}} = 2b\sqrt{a} \qquad a.e.$

and this algorithm also solves the problem to within 1.00 a.e.

2.3 Static, Multiple-Bus Problems

2.3.1 Introduction

We now consider the static problem in which there are k buses. The problem faced is essentially one of multiprocessing. Any passenger can be served by any bus and we are to allocate the passengers to buses in some optimal fashion.
Two possible generalizations of the previous criterion immediately arise, and we get the problems:

- (i) Minimize the total distance travelled by all k buses(i.e. minimize fuel used), and
- (ii) Minimize the time-to-completion (i.e. time-to-delivery of the final passenger).

A feasible solution to the problem is given by a set of k tours, one for each bus. For any such k tours, let

 x_n^i = distance travelled by bus i, i=1,...k

$$z_n^k = \sum_{i=1}^k x_n^i$$

 $y_n^k = \max\{x_n^i, i=1,...k\}.$

In (i) we wish to choose the tour to minimize z_n^k ; let Z_n^k be the optimal such value. In (ii) we wish to minimize y_n^k ; let Y_n^k be this optimal value. $(Z_n^k \text{ and } Y_n^k \text{ are random variables.})$

Note that

$$Z_n^{\dagger} = Y_n^{\dagger} = Y_n$$

 $(Y_n$ is as defined in the last section).

It is easy to see that for any instance of the problem we can always achieve

 $z_n^k \leq Y_n$

- simply use only one bus and keep k-1 idle. It appears reasonable to claim that $Z_n^k = Y_n$. This claim is in fact verified by Lemma 3.1.

Lemma 3.1⁽⁸⁾

 $\lim_{n \to \infty} \frac{Z_n^k}{\sqrt{n}} = c\sqrt{a} \qquad a.e.$ []

Thus, by increasing the number of buses we do not improve upon the total distance travelled or upon the fuel consumption. If instead, the final time-to-delivery of all passengers is of interest, we must investigate Y_n^k . Then we require that each of the buses absorbs part of the load, and the jobs of collecting and delivering passengers must be executed in parallel. For efficiency, all passengers should travel an equal distance, so that all buses are busy all the time. Clearly too, we must have

 $Y_n^k \geq \underline{Y_n}$

We ask, under what circumstances do we in fact get

 $Y_n^k = \frac{Y_n^1}{k}$?

Theorem 3 to follow indicates that this lower bound is attainable (in the limit) if passengers may transfer⁽⁹⁾ between buses, as in the scheme at Michigan. If passengers cannot transfer, so that the bus collecting a passenger must also deliver him, as in the scheme at Rochester, then each

(8) The proof of Lemma 3.1 which is given in the appendix is somewhat indirect, and an easier proof has been found to be evasive. However, the proof does identify additional problems as being of interest: these are problems in which at most i transfers are allowed during the tours of the buses, for i=1,2,3,...

(9) At a 'transfer point', at least two buses meet and the passengers on board are able to move arbitrarily between them.

bus must travel a greater distance.

2.3.2 Transfers Allowed

Theorem 3

If transfers are allowed

$$\lim_{n \to \infty} \frac{Y_n^k}{\sqrt{n}} = \frac{c}{\sqrt{a}} \qquad \text{a.e.}$$

Consider the following algorithm, producing a tour which we call $T_m^{k^O}$.

[]



Figure 2.2 Algorithm 3

Algorithm 3 (producing $T_m^{k^0}$)

Let m be any positive integer. Partition the region R into mk subregions each of area a/km. Bus i will be visiting only regions $r(i-1)m+1, \ldots r_{im}, i=1, \ldots k$. First, let each bus i visit region r(i-1)m+1, collecting all origins there. Then, let all buses meet at a prearranged transfer point (the point 0 in Figure 2.2) and passengers transfer onto the bus that will ultimately be visiting the region containing the required destination. Each bus i then visits region r(i-1)m+2, collecting all origins there and delivering all feasible passengers (from r(j-1)m+1, j=1,...k). The k buses meet at the next transfer point when passengers who have just been collected again transfer to their required buses. In this way, after m transfers, the buses will have collected all origins. Then (in the manner of Algorithm 2) bus i revisits regions r(i-1)m+1,...rim, delivering the remaining passengers. []

The m transfers enable a passenger to be on the bus visiting his destination as soon as possible. Whenever a bus enters a region at time t, it is able to visit every destination there that was feasible at t, irrespective of which bus visited the corresponding origin. Each bus visits 2n/k points and effectively serves n/k passengers. Furthermore, each bus travels through (1/k)th of the total area a (also, each passenger needs to travel at most through an area a/k) and travels a total distance $c\sqrt{n/k}\sqrt{a/k} = c\sqrt{n}\sqrt{a}/k$. The following corollary is then obvious (but see the detailed proof of Theorem 3).

Corollary 3

Given any $\varepsilon > 0$ there exist $N(\varepsilon)$ and $m(\varepsilon)$ such that for $n \ge N$ the tour $T_m^{k^0}$ given by Algorithm 3 is ε -optimal for problem (ii) (i.e. minimizing completion-time) with

- 34 -

probability 1.

For the tours T_m^o given by Algorithm 2, we defined a 'tour' $T^o = \lim_{m \to \infty} T_m^o$. There was no conceptual difficulty in doing this: the larger m became the sooner the destination whose origin had just been collected became feasible. The tour $T_m^{k^o}$ of Algorithm 3 now requires m transfer points; in the analogous limit as $m \to \infty$, a limiting optimal tour would require continuous transfers on the first passage through the region R. We thus refrain from defining a corresponding limiting 'tour'.

If we define the 'Michigan tour with k buses' to be the tour yielding $T_1^{k^0}$ (so this tour has only one transfer) what do we lose? As before, the distance travelled by each bus before the transfer (i.e. n/k origins in an area a/k are collected) is $b\sqrt{n/k}\sqrt{a/k}$. After the transfer, n/k destinations are visited in the same area. Hence, if $Y_n^{k^M}$ is the completion time which this algorithm yields, then

$$\lim_{n \to \infty} \frac{\mathbf{Y}_n^{\mathsf{M}}}{\sqrt{n}} = 2\mathbf{b}\sqrt{\mathbf{a}} \qquad \mathbf{a.e}$$

Theorem 3 gives that this tour solves problem (ii) to within (2b/k)/(c/k) = 1.06 a.e. once again; we lose but 6\$ by restricting ourselves to only one transfer.

-35-

[]

It is possible to improve on this algorithm slightly. Passengers with both origin and destination in the same area serviced by bus i, say, might in fact be able to be delivered before the transfer point. Thus, divide the region into mk subregions, with each bus servicing m regions, and a single transfer taking place after the first passage through these regions. Now m can be made large with no conceptual difficulty⁽¹⁰⁾. Such schemes will be discussed in greater detail for the dynamic problem of Section 3.3.

2.3.3 No Transfers Allowed

We now turn our attention to schemes in which transfers are not allowed. Of course, Lemma 3.1 still nolds, but Theorem 3 does not. It is easy to see that

$$r_n^k \ge \frac{r_n^1}{k}$$
 (2.6)

and, for k>1, it appears that a strict inequality holds if there are to be no transfers. We do not have results as strong as Theorem 3 for this case.

(10) This scheme yields a completion-time, $Y_n^{kM'}$ satisfying

 $\lim_{n \to \infty} \frac{Y_n^{kM'}}{\sqrt{n}} = \frac{2b\sqrt{a}[(1 + 1/k)^{3/2} - (1 - 1/k)^{3/2}]}{3} \quad \text{a.e.}$

The proof of this is similar to that of Theorem 6 (Appendix III).

- 36 -

(1) Two-Bus Case

For k=2 consider the following algorithm, which is a 'fixed-route scheme' for two buses (recall the fixed-route algorithm with one bus). Once more divide R into m subregions $r_1, r_2, \ldots r_m$. Let bus I visit the regions in order $(r_1, r_2, \ldots r_m)$ collecting and delivering all passengers with demands (r_1, r_j) , j>i. Let bus II visit the regions in order $(r_m, r_{m-1}, \ldots r_1)$, collecting and delivering passengers with demands (r_1, r_j) , j<i. Passengers with demands (r_1, r_1) are divided evenly between the buses and, as we can let m> ∞ these passengers will not cause difficulties. (See Figure 2.3.)



Figure 2.3 Two-bus static scheme without transfers

Each bus will visit n/m points in each region as before: for example, bus I in region r_1 will collect $(m-1)n/m^2$ origins (r_1,r_j) , j>1, $n/2m^2$ origins

-37-

 (r_1,r_1) and will deliver $n/2m^2$ destinations (r_1,r_1) . Thus, if Y_n^{2F} is the completion-time which this

algorithm yields, then

$$\lim_{n \to \infty} \frac{Y_n^{2F}}{\sqrt{n}} = mb \prod_{m=1}^{1} \prod_{m=1}^{a}$$
$$= b\sqrt{a} \qquad a.e.$$

Having considered numerous other two-bus schemes, we are lead to conjecture that

 $\lim_{n \to \infty} \frac{Y_n^2}{\sqrt{n}} = b\sqrt{a} \qquad a.e.$

i.e. that this scheme is asymptotically optimal (a.e.) for the minimal completion-time problem (problem (ii)) with no transfers.

Be this as it may, from (2.6) we obtain $\lim_{n \to \infty} \frac{Y_n^2}{\sqrt{n}} \ge \lim_{n \to \infty} \frac{Y_n^1}{2\sqrt{n}}$ $= \frac{c\sqrt{a}}{2}$ a.e.

so that the fixed-route scheme at worst solves problem (ii) with no transfers to within b/(c/2) = 1.0b a.e.

(11) Three-Bus Case

A similar scheme which we conjecture to be optimal for k=3 buses is the following. (See Figure 2.4.)

-38-



Figure 2.4 Three-bus static scheme without transfers

Divide R into 3m subregions r_1, r_2, \ldots, r_{3m} . Let $R_1 = \bigcup_{i=1}^{m} R_2 = \bigcup_{i=m+1}^{2m} R_3 = \bigcup_{i=2m+1}^{3m} r_1.$ Then bus I iservices demands $(R_1, R_1), (R_1, R_2)$ and (R_2, R_1) by travelling in order through (r_1, r_2, \ldots, r_m) , through $(r_{2m}, r_{2m-1}, \ldots, r_{m+1})$ and through (r_1, r_2, \ldots, r_m) again. (Call these the first, second and third passages respectively.) On the first passage the bus collects in $r_1, 1 \leq i \leq m$, all origins of the form $(r_1, r_j), i \leq j \leq 2m$, and delivers all feasible destinations, i.e. of the form $(r_j, r_1), 1 \leq j \leq i$ (in reverse order!) origins of the form $(r_j, r_1), 1 \leq j \leq m$ and $i < j \leq 2m$ are visited. Finally, on the third passage, all remaining destinations in $r_1, 1 \leq i \leq m$ are visited, i.e. $(r_j, r_1), m+1 \leq j \leq 2m$ and (r_1, r_1) . At the same time, bus II services demands (R_2, R_2) , (R_2, R_3) and (R_3, R_2) , and bus III services demands (R_3, R_3) , (R_3, R_1) and (R_1, R_3) , in a similar way.

As $m \rightarrow \infty$, the distance travelled by each bus, defined to be Y_n^{3F} satisfies

 $\lim_{n \to \infty} \frac{Y_n^{3F}}{\sqrt{n}} = .798b\sqrt{a} \qquad a.e.$

We derive this fact in the appendix, in Proposition 3.2.

(iii) k-Bus Case

The approach taken above can be used to obtain what appear to be good schemes for k buses, with k large.

Suppose there are $k=s^2$ buses. Divide R into s subregions and let each bus service a particular demand-type, i.e. (r_i,r_j) , i,j=1,2,...s. For $i\neq j$, the bus serving (r_i,r_j) will collect n/s^2 points in each of the regions r_i and r_j , yielding a time-to-delivery of

 $\frac{2b n a}{s^2 s} = \frac{2b\sqrt{n}\sqrt{a}}{k^{3/4}}$ (asymptotically a.e.) (2.7)

The buses serving demands (r_i, r_i) would complete their tours in less time, so (2.7) is the maximum time-to-delivery.

If k=s(s-1), we are able to do better since it is then possible to have all buses busy all of the time. Again with s subregions in R, let each bus service a demand (r_1,r_1) , if j and 1/2(s-1) of each of the demands (r_1,r_1) and (ri,ri). Then in each of its regions, a bus will visit

$$\frac{1}{(s-1)}\frac{n}{s^2} + \frac{n}{s^2} = \frac{n}{s(s-1)}$$
 points.

The total distance travelled by each bus is then

$${}^{2b} \int_{s}^{\underline{a}} \frac{n}{|s(s-1)|} = {}^{2b} \left[\frac{2na}{k(\sqrt{1-4k}-1)} \right]^{1/2} \quad (a.a.e.) \quad (2.8)$$

For very large k the values in (2.7) and (2.8) are not of too much interest. However, it is believed that the values are then 'almost optimal': they indicate the minimal time-to-delivery as a function of k, for k large.

The schemes for the multi-bus problem that we have analysed are seen in retrospect to have a certain specialization of the buses in common. For the case in which transfers are allowed, the buses are restricted to specialist <u>regions</u>. For the case in which transfers are not allowed, the buses specialize in collecting <u>demands</u> of a certain type only.

-41-

CHAPTER III

THE AVERAGE FLOW-TIME CRITERION AND THE DYNAMIC CASE

Two inadequacies of the models in Chapter II are the distance criterion, which is not really relevant to an individual passenger, and the static nature as, in actuality, we must consider an ongoing process, with demands arriving over time.

This chapter investigates these aspects of the problem by generalizing the previous results. In the dynamic case we shall consider the average flow-time criterion, so Section 3.1 investigates this criterion for the static case. Then, Section 3.2 analyses the (single-bus) dynamic problem and its relationship with the static problem. Finally, Sections 3.3 and 3.4 study the multiple-bus dynamic problem.

Once again, details of proofs are collected in an appendix. The notation and assumptions are the same as those described in Section 2.0.

3.1 The Static. Single-Bus. Average Flow-Time Problem

The criterion used in Chapter II - that of minimizing the time-to-delivery of the final passenger - is not necessarily satisfactory for an individual passenger. For,

SEDENG PAGE BLANK-NOT FILMED

he is interested rather in his personal wait-time and travel-time. Therefore, consider minimizing the average flow-time, i.e. the average time-to-delivery, of all passengers. This criterion will be used in the dynamic problem; the present section is needed as a link between the static and dynamic cases.

We need some additional notation. For any given problem instance of size n, let

 $0_n = set of n origins$

 $D_n = set of n$ destinations.

Any feasible tour T defines

Let

 $t(o_1|T) = time at which origin <math>o_1$ is visited $t(d_1|T) = time at which destination <math>d_1$ is visited.

 $w_n(T) = \underbrace{1}_{n} \underbrace{\sum_{i=1}^{n} t(o_i | T)}_{n i=1} = \text{average waiting time of } T$ $f_n(T) = \underbrace{1}_{n} \underbrace{\sum_{i=1}^{n} t(d_i | T)}_{n i=1} = \text{average flow-time for all}$ destinations of T.

The problem is, then, given a set of n demand pairs distributed at random in the region R with area a, minimize $f_n(T)$, with respect to all feasible tours.

We shall be solving a simpler version of this problem: essentially, we discretize the tours by considering only simple tours in S, those that visit the points in R region-by-region⁽¹⁾. Then, decisions are made at discrete instants of time - whenever a bus exits from a region. The choice of a tour corresponds to the choice of a strategy; at each stage that a bus exits from a region we must decide:-

(i) which region is to be visited next,

(ii) which points are to be visited in that region.

The restricted problem is

minimize $\{f_n(T), Tes\}$ (3.1)

In order to provide a Beardwood-type result for problem (3.1) we first consider a problem with an easier solution: minimize $\{w_n(T) + f_n(T), TeS_m\}$ (3.2)

We draw on classical scheduling theory for our informal discussion. Recall the following result (see e.g. Conway [6]). Given a single machine and n jobs to be processed, the average flow-time is minimized if the jobs are ordered in increasing processing-time order.

Suppose that each of the m subregions has area a/m, and that n_1 points are visited in a certain region. Then, the time spent there is, for large n_1 , approximately

It might seem that the restriction to S is a stringent one. However, for the minimum distance problems in Chapter II there is no loss of generality if we restrict ourselves to S, as we are interested in ε -optimal solutions: for any optimal tour T^e there exist an M and a T_M in S_M such that

 $Y(T_M) = Y(T^*) < \epsilon$ a.e. for n large enough. It is believed that this observation is valid for the average flow-time problem too, but a rigorous proof promises to be awkward.

⁽¹⁾ Recall the definition in Section 2.2.2.

 $b\sqrt{n_1}\sqrt{a/m}$, and we can take the time spent on each of the points to be $b\sqrt{a}/\sqrt{m}\sqrt{n_1}$. (Recall that the n_1 points are randomly distributed in the region and n_1 is large, so we can assume that the time spent between visitations is constant.) Now, to minimize the flow-time of all <u>points</u>, the scheduling result described above implies that we must visit first the points which will require least time, i.e. we must minimize $b\sqrt{a}/\sqrt{m}\sqrt{n_1}$. To do this, use the following rule.

Algorithm 4

Partition the region R into m subregions of equal area. At each stage, upon exiting from a region, choose as next region that with the most feasible points, and visit all feasible points there. []

This procedure corresponds precisely with the tour T_m^o given by Algorithm 2 (ties are broken arbitrarily). It minimizes the average flow-time of all <u>points</u>, i.e. origins and destinations, which is the criterion of problem (3.2).

Observe that the tour T° has a long average waiting time, $w_n(T^{\circ})$: passengers wait for collection while as many previous passengers as possible are delivered. Since T° minimizes $w_n(T) + f_n(T)$ in (3.2), we might expect that T° works well at minimizing $f_n(T)$ in problem (3.1). This is indeed the case, and T° is asymptotically optimal in (3.1) again. This is stated and proved (in the appendix) in Theorem 4.

-46-

More precisely, in Lemma 4.1 we derive the facts that, for the tour T_m° given by Algorithm 2, $\lim_{n \to \infty} \frac{w_n(T_m^{\circ})}{\sqrt{n}} = .575b\sqrt{a} + O(1/m)$ a.e. $\lim_{n \to \infty} \frac{f_n(T_m^{\circ})}{\sqrt{n}} = 1.142b\sqrt{a} + O(1/m)$ a.e.

Hence we have, from the considerations above,

Lemma 4.2

Let
$$w_n^+ + f_n^+ = \inf\{w_n(T) + f_n(T), Tes\}$$

Then,

 $\lim_{n \to \infty} \frac{w_n^+ + f_n^+}{\sqrt{n}} = 1.717 \text{b/a} \quad \text{a.e.}$

Further, given any $\varepsilon > 0$, there exists an $M(\varepsilon)$ such that for $m \ge M$ and n large enough, T_m^0 is ε -optimal for problem (3.2) a.e.⁽²⁾

Theorem 4

Let $f_n^* = \inf\{f_n(T), Tes\}.$

Then,

 $\lim_{n \to \infty} \frac{f_n^*}{\sqrt{n}} = 1.1417 \text{ b/a}$

Further, given any $\varepsilon > 0$, there exists an $M(\varepsilon)$ such that for $m \ge M$ and n large enough, T_m^0 is ε -optimal for problem (3.1) a.e. []

Thus, we have that the tour T° is nicely robust. It minimizes a cost relevant both to the passengers and to the operator of the system.

(2) Recall the footnote at Corollary 2. We require this stronger statement for the proof of Theorem 4.

It is worth observing that by combining Lemma 4.2 and Theorem 4, we obtain the result that the tour T^{O} minimizes with respect to TES any criterion

 $g.w_n(T) + f_n(T), g \in [0,1].$

As before, it is possible to compare the performance of suboptimal schemes. Let us look at tours in S_m , for fixed m.

It was seen earlier that the 'Michigan Scheme' corresponds to T_1^0 . The bus spends approximate time $b\sqrt{na}$ collecting the passengers and $b\sqrt{na}$ delivering them. The average time-to-delivery is then

 $f_n(T_1^0) = b\sqrt{na} + b\sqrt{na}/2$

= 1.5b/na.

Comparing with $f_n^{\#}$, we obtain that this tour solves the problem to within 1.5/1.1417 = 1.314, and we lose over 30% by restricting ourselves to tours in S₁.

Nonetheless, the sequence $\{f_n(T_m^o), m=1,2,...\}$ converges quite rapidly. For m=2, $f_n(T_2^o) = 1.330b\sqrt{na}$ and so T_2^o solves the problem to within 1.330/1.1417 = 1.165. For m=3, $f_n(T_3^o) = 1.269b\sqrt{na}$ solving the problem to within 1.112. Finally, for m=4, $f_n(T_4^o) = 1.238b\sqrt{na}$, solving the problem to within 1.08. Thus, by restricting ourselves to S₄ we lose only 8\$.

3.2 The Dynamic Case: Single-Bus Problem

Possibly the most glaring inadequacy of the previous problem formulations was the assumption that all demands were known at the starting time, t=0. For a more realistic formulation we must assume that the demands arise as time proceeds according to some intermittent arrival process. In this section we study the extension of the results from the static case to this dynamic version.

As before, we shall assume that any particular demand has both origin and destination drawn independently from a uniform probability distribution over the region R and that the demands are all independent of one another. For simplicity, we assume⁽³⁾ that demands arise at a constant rate, q.

To complete the problem specification we need a performance criterion. It is clear that, to minimize distance only without regard to time, is uninteresting: we can always wait until next year and then collect and deliver all demands together. So we choose to minimize the average flow-time. Thus, if a particular demand arises at time t_1 and the destination is visited at time t_2 , the flow-time for this demand is t_2-t_1 ; this is averaged over all demands.

(3) This assumption is unnecessarily harsh, and certainly we could assume a Poisson arrival process with mean q. However, the assumption is consistent with our informal presentation and it eliminates the need for many 'expectations'.

-49-

As in Section 3.1 we discretize the problem by restricting ourselves to the class of simple tours, S.

A standard easy heuristic for solving the dynamic version of a decision problem is to use the solution of the static version recursively - this loosely corresponds to the 'open-loop-optimal feedback' control scheme described by Dreyfus [7]. For our problem the technique is the following. At each stage compute the optimal (minimal flow-time) static tour for visiting all remaining points, assuming that no new demands will arise. Use this tour to yield the initial decision for the dynamic problem - i.e. which region to visit next, and which points there. By the time this has been implemented, new demands will have arisen. With this new initial state, recompute an optimal static solution.

In general, the heuristic is suboptimal as it does not take into account that future demands will arise. But, consider its application to our problem. We claim that the static minimum flow-time problem which is faced at any particular stage is solved by the same principle given in Algorithm 4: at each stage visit the region with most feasible points and visit all such feasible points there. We do not prove this claim; a detailed proof would be as long as, and more involved than, the previous proofs, and no additional insights would be gained.

-50-

The following is a translation of the heuristic into a proposed solution for the dynamic problem - Algorithm 4 is applied recursively. We call the tour T_m^o ; it is believed that no confusion with the static tour can arise.

<u>Algorithm 5</u> (yielding the tour T_m^o)

Partition the region R into m subregions of equal area. In each subregion which is visited, perform a travelling salesman tour on all reasible points. Upon exiting from a subregion, choose as the next one that with most feasible points. []

In order that the proposed static tour at each stage be (almost) optimal, the arrival rate q must be very large. Then the tour T_m^0 will visit the regions consecutively. For, with probability 1, the region with the most feasible points will also be the one with longest elapsed time since it was last visited.

How well does Algorithm 5 perform? Using a fundamental result from queuing theory we argue the following lemma in the appendix.

Lemma 5.1

If the system is in steady-state, then with probability 1, the tour T_m^0 given by Algorithm 5 minimizes, among tours in S_m , the average flow-time of all passengers. [] With this knowledge it is possible to give a Beardwood-type relationship for the dynamic problem when it has reached an equilibrium. Since q is the arrival rate of passengers, in order that the system be in a stable steady-state, we must serve q passengers per unit time. Thus, on average, 2q points must be visited per unit time. For the tour T_m° in S_m described above, let

 θ_q = the time spent in each subregion when the arrival rate is q.

During this time, $2q\theta_q$ points must be visited; so by Theorem 1 (recall that the buses travel at unit speed),

i.e. $\lim_{q \to \infty} \frac{\theta_q}{\sqrt{2q\theta_q}} = b \boxed{\underline{a}} \qquad \text{a.e.}$ $\lim_{q \to \infty} \frac{\theta_q}{\sqrt{2q\theta_q}} = 2b^2 \underline{a} \qquad \text{a.e.} \qquad (3.3)$

Define $P_q = m\theta_q$ as the 'period' - this is the time required for the bus to perform a circuit around R, visiting all subregions. Then,

 $\lim_{q \to \infty} \frac{P_q}{q} = 2b^2 a \qquad a.e. \qquad (3.4)$

From this it is easy to derive the formula below (see the appendix).

Theorem 5

Let $F_{q,m}$ be the optimal average flow-time for tours in S_m when the arrival rate of passengers is q. Then, $\lim_{q \to \infty} \frac{F_{q,m}}{q} = 2b^2 \dot{a}(1+1/m)$ a.e. (3.5) Hence, if F_q is the optimal average flow-time for tours in S, then,

$$\lim_{q \to \infty} \frac{Fq}{q} = 2b^2 a \qquad a.e. \qquad (3.6)$$

This theorem is analogous to Theorem 4, but for the dynamic problem. It is trivial that, by restricting ourselves to S_m instead of to S, we solve the minimum flow-time problem to within (1+1/m) a.e.

3.3 Dynamic Multiple-Bus Problem: Transfers allowed

Now let us extend the last section to the k-bus problem.

To introduce the approach, consider a 'Dynamic Michigan Scheme', a tour which we call (again) $T_1^{k^O}$.

Algorithm 6.1 (yielding $T_1^{k^o}$)

Partition the region R into k subregions, each of area a/k. Let all buses meet at a common transfer point⁽⁴⁾ at times $0, 20, 30, \ldots$ During time [i0, (i+1)0] each bus visits its own subregion collecting passengers who arrived during [(i-1)0,0] and delivering passengers who arrived during [(i-2)0, (i-1)0]. At the transfer point each passenger transfers onto the bus serving his required destination. []

(4) Recall Section 2.3.2 where this was justified by Figure 2.2. We omit any mention here of a 'line-haul' bus.

-53-

Here, $\theta = \theta_q$, the steady-state time spent by each bus in its region. During this time $q\theta_q$ additional demands arise with $2q\theta_q/k$ new points in each region. Then, as before,

 $\lim_{q \to \infty} \frac{\theta_q}{\sqrt{2q\theta_q/k}} = b \boxed{\frac{a}{k}} \qquad a.e.$ i.e. $\lim_{q \to \infty} \frac{\theta_q}{q} = \frac{2b^2 a}{k^2} \qquad a.e.$ The period P_q equals θ_q , so $\lim_{q \to \infty} \frac{P_q}{q} = \frac{2b^2 a}{k^2} \qquad a.e.$

Now seek the average time-to-delivery. Consider a passenger arriving during time $[0,\theta_q]$ where 0 and θ_q are transfer times. This passenger is collected during $[\theta_q, 2\theta_q]$ and is delivered during $[2\theta_q, 3\theta_q]$. The average time-of-arrival is $\theta_q/2$, the pick-up time is on average $3\theta_q/2$ and the delivery-time is on average $5\theta_q/2$ (the expected wait-time is θ_q , as is the expected travel-time). So the average flow-time by this algorithm, viz. $F_{q,k}^{M}$, is $2\theta_q$ satisfying

$$\lim_{q \to \infty} \frac{F_{q,k}^{M}}{q} = \frac{4b^2 a}{b^2} \quad \text{a.e.} \quad (3.7)$$

With k=1 this tour is precisely T_1^0 and thus (3.7) is but equation (3.5) with m=1, as it should be.

The tour $T_1^{k^0}$ is in S_k . A demand arising during $[0, \theta_q]$ must wait until the next period before it can be collected. This wait-time can be improved by further dividing the region of each bus (with area a/k) into m

equal subregions of area a/km each. Let each bus visit all of its m subregions successively, visiting as many origins as possible in each, and all the feasible destinations - now 'feasible' refers to the fact that the corresponding origin has been visited and the passenger is on the bus. After visiting their m subregions, the buses all meet at the same transfer point as before, passengers transfer, and the cycle is repeated. We call the resulting tour $T_{1,m}^k$; it is in S_{mk} and it requires a single transfer per period. Compared with T_1^{kO} , the waiting-time of passengers is reduced and passengers with origin and destination both served by the same bus might not have to visit the transfer point. It can be shown (this is a special case of Theorem 6 to follow) that this flow-time, viz. $F_{G:1,m}^k$, satisfies

 $\lim_{q \to \infty} \frac{F_{q;1,m}^{k}}{q} = \frac{b^{2}qa}{k^{2}} \begin{bmatrix} 3 - \frac{1}{k} + \frac{1+k}{mk} \end{bmatrix} \quad a.e. \quad (3.8)$

With k=1, (3.8) corresponds to (3.6) as it should.

In what sense are the two schemes $T_1^{k^0}$ and $T_{1,m}^k$ optimal? The answer might be forseen from the results of the static case and Theorem 5. An optimal tour in S_{mk} would require m transfers per period, one at each time that the buses (simultaneously) complete tours in their subregions of area a/km. This would ensure that a passenger who has been collected is transferred as soon as possible onto the bus that visits his destination. As a special case, T_1^k is optimal in S_k .

- 55 -

But, as pointed out earlier, it may be of interest to limit the number of transfers per period. Suppose we insist on only one transfer per period, and consider tours in S_{mk} . Then $T_{1,m}^{k^O}$ is optimal!

More generally, consider the k-bus problem in which we have at most h transfers per period, and we allow tours in S_{khm} . Then we have the following algorithm yielding the tour $T_{h,m}^k$.

Algorithm 6.2 (producing Th.m)

Divide the region R into knm subregions of equal area. Designate hm subregions for each of the k buses and label them $r_1^i, r_2^i, \ldots r_{hm}^i$, i=1,...k. Let bus i visit successively its regions $(r_1^i, r_2^i, \ldots r_m^i)$. Following this, all buses meet at a preassigned transfer point. Then bus i visits regions $(r_{m+1}^i, \ldots r_{2m}^i)$ and a second transfer occurs. After h such transfers, the buses again visit regions $(r_{1}^i, \ldots r_m^i)$ and the cycle repeats.

Every time a bus visits a region it visits as many points as possible in that region; these are all the uncollected origins and all the feasible destinations of passengers already on board. []

Theorem 6

The tour $T_{h,m}^k$ is optimal (asymptotically a.e.) for the k-bus minimal flow-time problem when tours are restricted to S_{khm} , and only h transfers are allowed per period.

Let $F_{q;h,m}^{\kappa}$ be the average flow-time resulting from this tour. Then,

1

$$\lim_{q \to \infty} \frac{F_{q;h,m}^{k}}{q} = \frac{b^{2}a}{k^{2}} \begin{bmatrix} 2 + \frac{1}{n}(1+1/k) + \frac{1}{n}(1-1/k) \\ mh & h \end{bmatrix} \quad a.e. \quad (3.9)$$
[]

From (3.6), when there is a single bus, the optimal average flow-time is $2b^2aq$. From (3.9), when there are k buses, the conceptual optimal tour in $S = S_{00}$ with continuous transfers (i.e. $h \rightarrow \infty$) has a flow-time of $2b^2aq/k^2$, and we have a k^2 -fold improvement. This is analogous to the static case⁽⁵⁾; each of the k buses visits (1/k)th of the total area a and has an effective arrival rate of q/k.

Using Equation (3.9) it is possible to investigate the optimality of suggested algorithms with respect to any restricted problem formulation.

For example, among tours in which only a single transfer is allowed per period, what is lost by using the tour T_1^0 ? Well,

$$f_q(T_1^0) = \frac{4b^2 a q}{b^2}$$
 a.

Letting $m \to \infty$ with h=1 we get from (3.9) that the optimal tour mas a flow-time of $b^2 aq(3-1/k)/k^2$ and so T_1^0 solves the problem to within

(5) In the static case in Section 2.3.2, the total distance travelled was (asymptotically) proportional to $\sqrt{n}\sqrt{a}$, yielding a k-fold improvement.

$$\frac{4}{3-1/k} = 1 + \frac{1}{3-1/k} e [4/3,2]$$

and we lose at least $33\frac{1}{3}$ ^{\$}. This is considerably more than the loss of 6^{\$} obtained for the static problem. For the dynamic problem, then, more is to be gained by having m large.

Or, what can be gained if we allow two transfers per period instead of one? With a single transfer we have a flow-time $(m \rightarrow \infty)$ of $b^2 aq(3-1/k)/k^2$. With two transfers the value is $b^2 aq(5/2 - 1/2k)/k^2$. The gain is

$$\frac{1 - \frac{5/2 - 1/2k}{3 - 1/k} = \frac{1 - 1/k}{2(3 - 1/k)} = 17$$

for k large.

These numerical values should not be taken too literally. But they do give qualitative insights into the relative optimality of suggested schemes.

3.4 Dynamic, Multiple-Bus Problem: Transfer-Free

Let us lastly consider some dynamic multiple bus schemes with no transfers. As in the static case of Section 2.3.3, we do not guarantee optimality. However, simple interesting schemes can still be studied.

-58-

3.4.1 Two-Bus Schemes

Consider the following scheme, T'(2), as shown in Figure 3.1. Partition R into m subregions. Bus I visits the subregions in order $(r_1, r_2, \ldots r_m, \ldots)$, and bus II visits them in order $(r_m, r_{m-1}, \ldots r_1, \ldots)$. Bus I serves demands of the form (r_i, r_j) where |(j-i)modm| < m/2, and bus II serves demands with |(j-i)modm| > m/2. Demands (r_i, r_i) are snared equally between the buses.



Figure 3.1 Two-bus dynamic scheme without transfer

For this scheme, let θ be the time spent by each bus in each of its subregions. During a time-interval θ , $q\theta$ demands (i.e. 2q θ points) arise. In equilibrium, each bus must collect $q\theta$ points in each subregion, i.e. $\theta = b\sqrt{q\theta}\sqrt{a/m}$, and $\theta = b^2qa/m$. The time for a bus to make a circuit of the m subregions is $P = m\theta = b^2qa$. For m

- 59 -

large, the average wait-time is P/2 and the average travel-time is P/4. Thus, for m large, the average flow-time for T'(2), viz. $F_q^{'(2)}$, is

 $F_q'(2) = \frac{3b^2}{4}qa$ asymptotically a.e.

A second two-bus scheme T''(2) is similar to that above, but with both buses travelling through the m subregions in the same order $(r_1, r_2, \ldots r_m, \ldots)$. Each bus always visits as many points as possible in each subregion it enters. We can suppose too that the buses are staggered, with bus II in region $r_{(i+m/2)modm}$ when bus I is in region r_i . For this scheme we have that, when m is large, the average travel-time is P/2 and the average wait-time is P/4, where $P = b^2qa$ again. Thus, if the average flow-time is $F_q^{"(2)}$, then for m large, $F_q^{"(2)} = 3b^2qa$ asymptotically a.e.

Although both schemes T' and T" have the same average flow-time, T' may be more desirable for a number of reasons. First, for T' the average wait-time is longer and the average travel-time shorter than those of T". So, with T', passengers have more 'coffee time at home'. Second, with T" the buses will not always remain staggered because of additional disturbances and uncertain travel-times. In fact, the buses will tend to merge together (a similar phenomenon is described by Newell [15]) so additional controls, causing additional delays, will be needed. Third, suppose that R is itself a 'subregion' within a larger system, and with a transfer point just before r_1 , say. Then T" is unfair in that a passenger originating in r_1 will always have a longer travel-time than one originating in r_m . The variance of the travel-times will be smaller for T'.

3.4.2 A Three-Bus Scheme

Finally, consider a three-bus scheme T(3) that is a simple extension of T'(2) above. The region R is divided into 3m subregions, each with area a/3m. The buses perform routes as shown in Figure 3.2; a passenger is served by the most suitable bus, that is, one that will provide a shortest travel-time for him. Let θ be the time spent by a bus in each of its subregions. During this time 2q0 points arise, and for equilibrium 2q0/3 points must be visited by the bus. So, $\theta = b\sqrt{2q0/3}/a/3m$ and $\theta = 2b^2qa/9m$ asymptotically a.e. Each bus has period $P = 2m\theta = 4b^2qa/9$. When m is large, the average wait-time for a passenger is P/2. The average travel-time is clearly less than this, and can be shown to be P/4. Hence the average flow-time $F_q^{(3)}$ is 3P/4, i.e. when m is large.

 $F_q^{(3)} = \frac{b^2 qa}{3}$ asymptotically a.e.

-61-



We conjecture that this is the minimal average flow-time for the three-bus transfer-free case. However, note again that a passenger requiring service from r_1 to r_{2m} has a very long travel-time even though the regions are adjacent. A six-bus scheme might have two buses travelling in opposite directions on each of the routes shown in Figure 3.2, and this would reduce the discrimination.

CHAPTER IV

TOWARDS THE PRACTICAL DESIGN OF DIAL-A-RIDE SYSTEMS

4.1 Introduction

In Chapters II and III we have presented a mathematical theory for idealized bus problems, which we have tacitly claimed is relevant to the real dial-a-ride problem. In order to justify this claim we must relate the real and idealized problems and draw upon the theory to obtain implications for real design. In fact, we would like to go even further and propose an approach for dial-a-ride scheduling, which is of both theoretical and practical interest. It is this that is the goal of the present chapter.

If a dial-a-ride system is required in a certain area, how should the service be designed and scheduled? Our response is essentially described by the following three planning steps. First, at the most basic level, it must be decided what approach is to be taken. The proposed approach is through the decentralization of the large problem: each bus is required to specialize only in passengers of a certain 'type', determined by their origin-destination pair. Once this approach has been adopted, the second aspect of design is that of designating the decomposition; to do this

65

FRECEDING PAGE BLANK-NOT FILMED

we must specify the roles to be played by the individual buses, we must select subregions and plan transfers. At the third level is the more detailed operational design; here, we must establish decision rules that will determine the behaviour of the buses when passengers arrive.

This trinity yields a class of scheduling algorithms which we wish to propose. The class is described essentially by five 'elements' in Section 4.2. We draw upon the theory of Chapters II and III to motivate these qualitative elements and also to derive quantitative techniques for design. In Section 4.2 too, some of the theoretical assumptions and their limitations emerge more clearly. Adjustments are therefore required if the quantitative approach is to be practically meaningful, and some of these are discussed in Section 4.3. Finally, to give an improved grip on the ideas, we describe an example in Section 4.4, a 'pseudo-real situation', in which the approach is applied.

4.2 A Design Approach

This section describes, in some generality, an approach towards the design of dial-a-ride scheduling algorithms with which we shall be concerned. Given are a region where the transportation system is to be designed, and an anticipated probability distribution according to which passenger

-66-

demands will arise. A certain scheduling facility (a computer?), receiving these demands, is to determine suitable tours for a fleet of buses.

The following are five elements that characterise a class of dial-a-ride scheduling algorithms:

(i) Decomposition of Area and Specialization of Buses

(ii) Simple Tours

(iii) Rules for Visiting Subregions

(iv) Transfer Points

(v) Method of Quotation.

Our interest in these algorithms is motivated by the models of Chapters II and III. We shall discuss the five elements in detail and draw upon the theory to explain and to justify them. For this, it must be assumed that the idealized models do indeed focus upon the quintessentials of the real problem. Of course, for tractability, many assumptions and simplifications have had to be made; some of the more important of these will be discussed in Section 4.3. Despite the assumptions, the many reasonable qualitative implications do yield interesting algorithms. And the algorithms, when extended to the real problem, have many features of practical interest to boot.

Undoubtedly, fundamental to the idealized algorithms in Chapters II and III is their decomposed and discrete nature: the buses specialize in certain demand-types only, and the tours are all 'simple'. For example, recall Theorems 2 and

120

3 and the tour T_m^0 which approximate to any required accuracy the optimal performance⁽¹⁾. Even very simple-minded algorithms of this form yield good performance. Thus we are led to claim that the following two elements of design are worthwhile.

(i) Decomposition of Area and Specialization of Buses

Initially, the region in which the transportation system is to be constructed, is partitioned into a large number, m, of subregions. Then, a passenger requesting service from a subregion ri to a subregion r; can be directly classified according to his demand-type, (r_i, r_j) - there are m² such demand-types. Now, assign to each of the buses a fixed set of subregions which it visits. Each bus is to specialize within its subregions in serving certain demand-types only. That is, it visits only their origins, or destinations, or both. A bus may also periodically visit 'transfer points'. At these, it collects additional passengers and it delivers a subset of its current passengers, each of whom will be collected by some other bus.

(1) The measure of performance is distance travelled or average flow-time. The schemes have thus been found to be robust, minimizing criteria relevant both to the operator of the system and to the passengers. We shall discuss these, and other measures of performance, subsequently.
By the discussion in Section 2.3, we should assign buses to the demands in such a way that all buses share an equal portion of the 'load' and that each bus is busy all the time. We should also ensure that the buses are always suitably dispersed over the area: this will promise good service to the future demands that arise according to the given probability distribution.

The number m of subregions to which each bus is assigned is important, particularly in dynamic problems. However, this number does not need to be too large before good performance is attained (see Sections 3.1 and 3.2). When the demand distribution is uniform, the subregions can all simply be of equal area, yet otherwise arbitrary⁽²⁾. In practice, however, physical constraints such as boundaries of towns, major roads, and so on, will influence this choice.

Once the buses have specialized, the demand-type of a passenger who requests service will trivially determine the bus, or sequence of buses, to which he is to be assigned. The central scheduler that receives the demand performs no long search, but only informs the relevant bus and provides the book-keeping.

(11) Simple Tours

Passengers' demands arise over time. Depending

(2) Additional remarks will be made in Section 4.3.

-69-

upon the actual requests, tours for the buses within their subregions must be determined. Any point which a bus visits must be 'feasible' - it must be an origin or a destination of a passenger who is aiready aboard. (These points are of course within the bus's speciality.) The points are to be visited in some 'optimal' order. The problem is simplified by constructing tours in discrete stages, whenever a bus enters a new subregion. At this time of entry, let the bus calculate an optimal travelling salesman tour in the upcoming subregion; any point visited there is feasible at this instant of time. The calculation is to be done on-line.

We have termed tours of this form 'simple'. Recall that the use of these tours was justified for the static problems of Chapter II, and it was claimed that they were suitable for the dynamic problems too (Chapter III, page 43, footnote (1)).

If the subregions are small and only a few points are to be visited in each, it may be easy for a driver to 'see' what route he should take. Otherwise, he might have a minicomputer aboard that finds a tour for him. Efficient approximation algorithms for the travelling salesman problem are given in [12,3]. In this way, the calculation of tours is decentralized - it reduces to a number of small discrete

-70-

problems. By choosing the number of subregions carefully, it is possible to guarantee that the total computation time is bounded by a polynomial function of q, the arrival-rate of passengers (see Karp [9]). Note too that the communication between the buses and the central scheduler does not need to be continuous, so relatively cheap communication equipment should suffice.

(iii) Rules for Visiting Subregions

Two questions remain whenever a bus exits from a subregion at a time t_1 : which region should it enter next, and which points should it visit there? Well, in any region it enters, it must visit all points within its specialization that were feasible there at time t_1 . And, it should visit its regions successively⁽³⁾, in a fixed order.

These rules were proved valid for the case in which demands were uniformly distributed, in Section 3.1. Essentially, they require that buses 'always do as much as is possible'. They imply that a bus will not deviate in a backward direction from its prospective route to service

(3) This should be further qualified. It holds when the distribution of points is uniform. In the case of more general distributions, a generalized rule may be required. For example, visit next the region with largest 'point density' (i.e. the number of feasible points per unit area) - see Section 4.3(i).

-71-

newly arriving demands, as this may delay future passengers. After t_i , feasible points that arrive in the subregion are not collected: a passenger who just misses the bus must wait for the next one to enter the subregion. Even though some passengers are inconvenienced, their contribution to the average performance is kept small.

It may happen that, in a particular subregion, there are no feasible points. Then, the bus can simply bypass that subregion or travel through it as efficiently as possible to the next. We can think of the bus as travelling through its subregions along a fixed route and deviating within each subregion to collect and deliver passengers there. Passengers then know the direction in which the bus is proceeding, and should be happy, knowing roughly where they are headed.

(iv) Transfer Points

The transfer points (see (i) above) at which a bus stops from time-to-time are typically on the boundary of some subregions. They must be considered during the calculation of the optimal travelling salesman subtours. Our description has been sufficiently general to allow, for example, that some buses have 'null subregions', travelling only between transfer points; these are the familiar line-haul buses. Transfer points were shown to be necessary in Sections 2.3.2 and 3.3 if the delay incurred can be neglected. For, they provide vast improvements in performance, and this improvement increases with the number of transfers. However, too large a number of transfer points is clearly impractical, and passengers might attach an additional cost to the number of transfers they have to make. Also, the theoretical approach of having all buses meet at the same transfer point may not be desirable:- the additional trip from the region to the transfer point will not be negligible when the number of passengers is small. In this case, the line-haul bus effectively creates the single transfer in a practical way.

(v) Quotation Times

In the statement of the dial-a-ride problem in Section 1.1 it was required that, at the time a passenger requests service, he be quoted a time of collection and delivery. The scheduling procedure we have described has not needed such a quotation. We regard it as a separate secondary problem, requiring that an estimate of the 'load' on the system, and hence an estimate of the speed of the buses, be made.

-73-

The approach which we have described guarantees a certain average service to the set of all passengers. Thus, if quotations are made to individual passengers, and if these are then introduced as additional constraints, the average service to passengers will be reduced. In practice, quotations are required, but the scheduler need not be committed to them. We believe that simple constant estimates of the times of collection and delivery will usually suffice. It is known through which subregions a bus is to travel: the time spent deviating in each subregion could be estimated and quotation times could easily be provided. So, an accurate prediction of travel-time between points is avoided.

The elements (i)-(v) above outline, in somewhat general terms, a class of scheduling algorithms. The basic approach is described by (i) and (ii) and we have claimed that this approach is justified in an asymptotic probabilistic sense. The detailed operational design is described by (iii) for a special case, and in the next section it will be seen that this can be extended fairly generally. What is lacking, then, is a methodology for selecting a suitable decomposition and for organizing the manner in which the buses specialize.

For the special-case idealized models of Chapters II and III no such methodology is required since the solution is relatively easy: when the demands are uniformly

-74-

distributed over the region, partition it into subregions of equal area. Then, in the manner of Algorithm 6.2 the buses specialize in disjoint subregions and have cyclic tours through them when the arrival-rate of passengers is large. In general, however, the answer is not as easy and difficulties, besides the nonuniformity of distributions, abound in practice. Indeed, let us remark upon a few practical considerations which indicate that a general methodology for the determination of subregions and bus-routes would surely be very complex.

First, what criteria actually determine good service? The criteria assumed in Chapters II and III, viz. the distance travelled by the buses and the average flow-time of passengers are not all-important, but are only first approximations to real requirements. There are, for the variance of passengers' travel-times example. (measuring, in a sense, the 'fairness' to the passengers) and the maximum of all passengers' travel-times (since we wish to provide all demand-types with good service). Or, there are questions of the reliability of the system (what happens if a bus breaks down?) and its stability (what effect will additional delays and uncertain travel-time have?).

Furthermore, a new dial-a-ride system would have to suit the existing circumstances. Boundaries of towns, rivers or roads may impose physical constraints on the way

-75-

in which the region can be subdivided. We may wish to supply a special quality of service to certain demand-types (e.g. to disabled persons, or see Example 4.4). A dial-a-ride system must be compatable with existing public transportation systems: it should supplement them, and yet could also depend upon them. For example, existing bus- or subway-routes might reduce the number of line-haul buses that are needed if the subregions are designed to take advantage of them.

Thus, dial-a-ride schemes must be tailor-made to suit individual complex requirements. Even though a general methodology has not been proposed, our study has yielded many qualitative insights, and no doubt additional principles could be obtained with further research.

On the other more favourable hand, however, we wish to suggest that our approach does yield a practical design tool. The emphasis throughout Chapters II and III was upon the comparison of algorithmic performance (see Definition (2.1)). In general, it is possible to quantify the performance of a suggested scheme whenever a distribution of demands, a decomposition of the region into subregions and an organization of specialist buses, are given. For, one can estimate the time spent by each bus in each of its subregions (using Beardwood's formula, Theorem 1) and evaluate its dynamic behaviour as it travels around its

-76-

subregions and visits its transfer points (4). The service supplied to the various demand-types by the suggested scheme can then be evaluated with respect to average flow-time and to many other performance criteria. Thus, a suggested scheme can be quantitatively studied. Usually, closed-form analytic formulas should be obtainable to express its Then, investigations can be made to characteristics. analyse the behaviour of the scheme when the parameters e.g. the number of buses, the demand distribution, the number of transfer points, and so on - are varied. Finally, alternative schemes, with alternative regional decomposition and bus-specialization can be evaluated and compared. These are analytic studies, and avoid the need for complex simulations. An example that briefly illustrates such a quantitative investigation is given in Section 4.4.

It is in order, finally, to make some additional remarks on the design approach which we have expounded in this section.

The approach can be applied on a broader scale. The 'city' may itself be a module, connected to others within a larger system. Also, the method is relevant to 'one-to-many' systems, [21], where passengers have a common origin and wish to be taken to various destinations. For, it is easy to assign incoming passengers to buses

(4) This was illustrated for the simple models in Chapter III.

-77-

simultaneously, merely treating the single origin as a transfer point. 'Many-to-one' systems are similar.

In [21], the existence of 'advance requests', where passengers request service some time in advance, is regarded as important. With our approach these demands are easy to handle. If the 'period' of a bus is P - this is the time it takes for the bus to make a circuit of all its subregions - the passenger can be promised collection during $[t_1,t_1+P]$ for any suitable time t_1 . Then, when the region is entered, the passenger will be collected if this can be during $[t_1,t_1+P]$, otherwise the collection is deferred. Delivery by a certain time can similarly be guaranteed in advance.

There is a further important implication of the decomposition described in (i) above. The success of a depends transportation system ultimately upon the co-operation of the bus-drivers, particularly if they have decision-making capabilities. Since each bus has certain fixed demand-types assigned to it, it is possible to measure the efficiency of individual drivers. If each driver travels as well as possible through his subregions, the total system performance will be maximized. So, some incentive might be given to the drivers that will encourage them to perform well and, for example, to reduce the number of unnecessary stops. Furthermore, since the drivers communicate with the central scheduler only intermittently,

- 78 -

the 'big-prother' paranoia of continuously-watched drivers is reduced.

4.3 Quantitative Aspects of Design

In Section 4.2 it was described how suggested schemes can be investigated by quantifying their behaviour. Investigations such as these are limited by the many theoretical assumptions of Chapters II and III, since inaccuracies are then produced in the resulting measurements. In this section we discuss some of the more important of these assumptions and investigate to what extent they can be relaxed. We indicate how the inaccuracies might be reduced by making adjustments in the applied formulas.

4.3.1 The Demand Distribution

In deriving formulas measuring, for example, the average flow-time for a suggested scheme (see Theorems 5 and 6) Theorem 1 is used to obtain the time spent by a bus in each of its subregions. For this, it must be assumed that the points which are visited in each subregion are drawn from a uniform probability distribution there: this was the case in our previous analyses since the demand pairs were 'random'. transportation system is to be designed, and an anticipated probability distribution according to which passenger

-80-

Suppose, now, that the points to be visited in a subregion are not uniformly distributed, but arise according to a probability density function, p, say. Then it has been shown in [2] that the optimal tour-length L_n satisfies

 $\lim_{n\to\infty}\frac{L_n}{\sqrt{n}} = b\int p^{1/2}dv \qquad \text{a.e.}$

When p = 1/a, we have the uniform distribution. Since $\int p^{1/2} dv \leq \sqrt{a}$, nonuniformity of the distribution decreases R L_n. Consequently, the additional structure in the nonuniform distribution can be used to advantage.

The expression above might be used to improve the analyses when the distribution is nonuniform, and when all origins and destinations are independent of one another. The discretization imposed by simple tours will remain useful, however. For, if the whole region is divided into m subregions and if we approximate the distribution p to be uniform within each subregion, then by making m large, the approximation can be made arbitrarily negligible.

Now, consider a bus visiting its subregions. What rule (i.e. element (iii)) should it use? It is not hard to see that in order to minimize average flow-time the bus should always 'do as much as possible' (compare with Lemma 5.1). Thus, whenever it enters a subregion it should always visit any points as possible there. But, which subregion it which subregions there are ni feasible points to be visited and if then $T_i(n_i)$ will be the time spent in each, choose i so as to minimize $T_i(n_i)/n_i$. Even when the arrival-rate q of demands is large, it may be the case that the bus does not visit its subregions successively. In general, the analytic calculation of average flow-times will be more difficult than before, but will still be feasible.

Note that these remarks are not true if the origin and destination in a demand pair are correlated.

Finally, it is in order to comment upon the stochastic process by which demands arrive over time. We assumed a constant arrival-rate q, and the analysis could be extended for a Poisson process with mean q. In practice, q will change over time. This change will be relatively slow - at certain times of the day alternative designs for the decomposition and specialization might be selected, so as to suit the changing operating conditions. It is believed that a few discrete such changes should suffice in practice.

4.3.2 The Metric

We have taken as the distance between any two points in the city their euclidean separation - i.e. the length of the straight line joining them. In reality, the distance travelled by a bus is not as simple. If the actual distance can be approximated by a constant multiple of the euclidean distance, the analysis of Chapters II and III remains valid

-81-

(1) The measure of performance is distance travelled or average flow-time. The schemes have thus been found to be robust, minimizing criteria relevant both to the operator of the system and to the passengers. We shall discuss these, and other measures of performance, subsequently.

-82-

with very minor modifications.

Alternatively, the distance between two points x, y may be better approximated by a multiple of the 'Manhattan metric', or L^1 , given by

distance(x,y) = $|x_1-y_1| + |x_2-y_2|$.

Again the results hold, this time with a different value of the constant b [2].

More general metrics that take into consideration indirect or faster roads, promise to be an order of magnitude more difficult: for, there is then no result corresponding to Theorem 1.

4.3.3 Asymptotic Approximation

Possibly the most glaring inadequacy of the approach is that it is asymptotic in the number of passengers. The expressions which are obtained are laws of large numbers, and require that the number of passengers and their arrival-rate q be large. To apply the formulas we would ideally seek a 'central limit' result allowing us to estimate the probable magnitude of the errors when q is small, and to determine the size of q needed for reliable estimates.

Sadly, a general result of this nature is not available. A theoretical analysis is hard and results depend, for example, upon the shape of the region. For R a square, the best that it has been possible to do analytically on the convergence of Theorem 1, is to show that (5)

$$\frac{E[L_n]}{\sqrt{n}} = b\sqrt{a} + O(\log n/n)$$

where $E[L_n]$ is the expected length of an optimal travelling salesman tour for a random instance of the problem of size n. It might be hoped that the actual rate of convergence is faster than this.

In this spirit, let us investigate empirically the convergence of Theorem 1. We shall consider only points which are distributed randomly in a unit square, and shall seek to estimate $E[L_n]$ as a function of n. Comparing $E[L_n]$ with $b\sqrt{n}$ will yield an expected error in the prediction. This error will enable us to bound, in a sense, the error due to the asymptotic approximation in dynamic versions of the problem as well.

The value of $E[L_n]$ was estimated for n=20,30,... as follows. Random travelling salesman problems were generated by scattering n points on the unit square (x and y coordinates were generated uniformly and independently). A number of instances of each problem size n were solved⁽⁶⁾.

(5) This can be derived in the manner of Beardwood's Lemma 6.

(6) It should be remarked that a similar experiment was performed by Christofides and Eilon [4]; however, details of their numerical results and the algorithms used were not provided.

-83-

are given in [12,3]. In this way, the calculation of tours is decentralized - it reduces to a number of small discrete

-84-

The algorithm used was a heuristic of Lin and Kernighan [12]: it uses modest computation time and has been found to 'practically guarantee optimality'. For n large the computation time was still high, however, and fewer problems were sampled. The results obtained are tabulated in Figure 4.1. From these, we estimate the value of b to be about .765; Figure 4.2 compares the empirical expected

tour-length with the value of $.765\sqrt{n}$. In Figure 4.3 we plot $E[L_n]/\sqrt{n}$ versus n, indicating how it converges to .765. The curve .765 + 4/n is given for comparison: it seems clear that $E[L_n]/\sqrt{n}$ converges at a faster rate than O(1/n).

It is interesting to note that for every random problem-instance tried, with b = .765,

 $b\sqrt{n} \leq L_n$.

So, with high probability the tour-length predicted by Theorem 1 is <u>less</u> than the actual value. Indeed, from Figure 4.3,

 $b\sqrt{n} \leq E[L_n] \leq b\sqrt{n} + 4/n$ (4.1) This yields a fairly close bound on the magnitude of the error implicit in Theorem 1.

Now let us obtain an estimate of the convergence rate for a bus problem - we are interested primarily in the dynamic version. In evaluating the performance of a scheme the time spent by a bus in each of its subregions is estimated, so we can utilize the results above. We consider For example, visit next the region with largest 'point density' (i.e. the number of feasible points per unit area) - see Section 4.3(1).

Random instances of the travelling salesman problem, generated on the unit square, Figure 4.1

-85-





only a very simple problem: a single bus is serving a region of unit area and visits m subregions successively. Each subregion is a square with area 1/m, and the system is in steady-state.

Let q be the arrival-rate of demands, and let $E[P_q]$ be the expected period - i.e. the time for the bus to visit all m subregions. In Equation (3.4) we predicted that $E[P_q]$ would behave asymptotically as $2b^2q$. We now ask the following simplified question: can we bound the relative error in this prediction by a function of q. That is, seek a bound to

error =
$$\left| \frac{E[P_a] - 2b^2q}{2b^2q} \right|$$
 (4.2)

Let us make another simplifying assumption. Suppose that, as the bus travels through its subregions, it visits the same number of points, n, in each⁽⁷⁾. Then, the expected time spent in each subregion is $E[L_n]\sqrt{1/m}$, and the expected period is $E[P_q] = mE[L_n]\sqrt{1/m}$. During a period, mn points are visited,

1.e.

 $q = \frac{n}{2E[L_n]\sqrt{1/m}}$

(4.3)

(7) This assumption drastically simplifies the analysis. The number of points to be visited in each subregion is really random and we get

Expected time in a subregion = $\sum_{n=0}^{\infty} p_q(n) E[L_n] \sqrt{1/m}$

where $p_q(n)$ is the probability that there are n points in the subregion when the arrival-rate is q. (Note that there is a correlation between the number of points n visited in successive subregions: there will be sequences of large n and sequences of low n.)

-88-

rivers or roads may impose physical constraints on the way

-89-

From (4.2),

error =
$$\left| \frac{E[L_n] - b^2 n/E[L_n]}{b^2 n/E[L_n]} \right|$$

Using (4.1),

error
$$\leq \left| \frac{(b\sqrt{n} + 4/n)^2 - b^2 n}{b^2 n} \right|$$

= $\left| \frac{8b\sqrt{n} + 1b/n}{b^2 n^2} \right|$

With b = .765, the relative error given by (4.2) is less than 12% when n, the number of points visited in each subregion, is 20; it is less than 7% when n is 30; and it is less than 5% when n is 35. For any n, (4.3) determines the corresponding arrival-rate, q.

4.3.4 Additional Uncertainties

The following are a few additional issues (mainly related to stochastic variations) that we have neglected and that we now wish to note.

(i) The speed of a bus is not constant as has been assumed. The travel-time between two points is uncertain and is subject to traffic and road conditions.

(ii) Delays that occur when a stop is made have been neglected. A bus may have to wait for a passenger to appear and for his impedimenta to be loaded. Similar delays occur for alighting passengers.

(111) A bus is also detained at transfer points. The time

taken is not instantaneous, but is a function of the number of passengers who transfer. Indeed, if buses are to meet at transfer points as was required in Chapters II and III, waits are inevitable.

(iv) From a passenger's viewpoint, the time spent on a line-haul bus (including his wait) may be appreciable. In our asymptotic analysis these, as well as the time taken for a bus to travel a fixed bounded distance were neglected.

(v) Consider a bus travelling a circuit around its subregions. In Chapter III we calculated the equilibrium value of the average flow-time. If, now, the bus is delayed, additional passengers still arrive, and the time spent in the following subregions will be longer than anticipated. So, the delay will persist and the equilibrium will only be re-achieved gradually.

Because of these assumptions, there will be yet further errors in the derived formulas. However, it is believed that many of them could be taken into account in a practical analysis. For example, studies of urban traffic behaviour might be used to more fully understand the travel-times; and we might be able to estimate the time spent by passengers on the line-haul system. Then, the tools which we have proposed could be further sharpened and calibrated.

-90-

(4) This was illustrated for the simple models in Chapter III.

-91-

Finally, a more sophisticated technique for the detailed operational control - one that will take into consideration the random fluctuations - can be suggested. Consider specifying in advance a schedule of times for the buses to be at transfer points. Then, if a bus is running late, it can improve its speed by selecting carefully which of its passengers it should collect: some passengers remain uncollected until the bus (or the next suitable bus) re-enters their subregion. This feature would require additional machinery if it is to be analysed. A measure of the 'state' of each bus would have to be defined, and it would be nontrivial to determine the schedule in advance certain 'slack' would be required to allow for random disturbances. The schedule would have to be 'stable' with respect to reasonable perturbations and 'flexible', so that changes can easily be made if things go awry. To study these topics would require new methods of some 'stochastic scheduling theory'.

4.4 An Example

Let us suppose that we have a city, R, in which we want to construct a dial-a-ride transportation system. This city is essentially composed of two distinct regions as shown in Figure 4.4 - downtown, R_d , with area a_d , and suburbs, R_s , with total area a_s . There is already an efficient public transportation system operating in R_d . of unnecessary stops. Furthermore, since the drivers communicate with the central scheduler only intermittently,



-92-

Figure 4.4 An example : the city

For most of the day there is heavy traffic travelling between R_s and R_d in both directions. The dial-a-ride scheme is to be designed for this scenario.

Suppose we can approximate the distribution of demands that are to be served by our system by the following. Any particular demand pair (o,d) is of the form (R_s,R_d) with probability p/2, pE(0,1); it is of the form (R_d,R_s) with probability p/2; and it is of the form (R_s,R_s) with probability (1-p). This distribution, then, is nonuniform. The system is not required to serve demands (R_d,R_d) . Further, the value of p is large, i.e. fairly close to unity.

As remarked, the approach which we wish to adopt is one in which R is decomposed into subregions and in which buses specialize. The theory of Chapters II and III has not 'random'.

-93-

given a general methodological procedure for designing this decomposition when demands are nonuniform, but the following could well be reasonable for the present example.

In order to discourage persons who travel between R_s and R_d from using their private vehicles, let us provide them with attractive transportation, say a transfer-free service. We describe first a scheme for only these (R_s,R_d) and (R_d,R_s) demands. Consider a division of R_s into k regions, $r_1,\ldots r_k$, each of area a_s/k , for an integer k. It might be desirable to have natural divisions here, described by boundaries of suburbs, roads and so on. This is depicted in Figure 4.5.



Figure 4.5 Partition of suburbs, Rs

Now allocate a bus b_i to the region r_i , for i=1,...k; bus b_i will serve passengers with origin or destination in r_i and will travel also in R_d . Each bus

AD-A046 162 HARVARD UNIV CAMBRIDGE MASS DIV OF ENGINEERING ANDETC F/G 12/2 SCHEDULING DIAL-A-RIDE TRANSPORTATION SYSTEMS: AN ASYMPTOTIC APETC(U) N00014-75-C-0648 NL														
	2 OF 2 AD 46162					÷.					An and a second			
														1994
		<u>¥</u>												A. Gittlet 12.1
					 M. BARTANI AND AND AND AND AND AND AND AND AND AND	scradingerts have been been been been been been been be	END DATE FILMED 12-77 DDC		All the Court	Boosting			Constant and a	,
		daa to			1	l.								
~		_		1.1.1							_			/

will travel a 'simple tour' (see Section 2.2.2), so subdivide r_i , i=1,...k and R_d further, into m subregions each. Bus bi travels a route as shown in Figure 4.6. It makes a circuit around its 2m subregions, visiting them successively. If it enters a subregion ri,j at a time t, it visits there all possible points that are feasible at t. These are the origins of demands that have arisen during the time since rij was last entered, and all destinations in ri,j of passengers who are already aboard bi. The tour in ri,j is an optimal travelling salesman tour, and can be calculated at t. If the number of points to be visited in ri, j is small, the driver might easily be able to choose his best tour. Otherwise, there might be a minicomputer aboard that calculates the tour for him. Efficient approximation algorithms that might be used for this are those in [12,3].



Figure 4.6 Route taken by a single bus in a region

-94-

The buses communicate with a central computer (or scheduler) that informs them of the points to be visited in their subregions. This computer monitors all demands and assigns them to the buses; the assignment decisions are trivial - no long or hard search is needed.

With k regions of R_s and k buses, each travelling at unit speed, the asymptotic equilibrium average flow-time for the (R_d,R_s) and (R_s,R_d) demands in the above scheme, can be calculated using the techniques developed in Chapter III as follows. Consider the route of a particular bus. Let θ be its period, the time for it to make a circuit of its 2m subregions⁽⁸⁾. Let θ_d be the time spent in each subregion of R_d , and let θ_s be the time spent in each

$$\mathbf{\Theta} = \mathbf{m}\mathbf{\Theta}_{\mathbf{S}} + \mathbf{m}\mathbf{\Theta}_{\mathbf{d}} \tag{4.5}$$

During θ , there arrive $2pq\theta/k$ points that are to be served by this bus, i.e. $pq\theta/km$ in each subregion. Thus, by Theorem 1.

$$\theta_d = b \frac{pq\theta}{k} \frac{a_d}{n}$$
(4.6)

 $\theta_s = b \frac{pq\theta}{k} \frac{a_s}{kn}$
(4.7)

From (4.5), (4.6) and (4.7),

 $\theta = \frac{b^2 pq}{k} \left[\sqrt{ad} + \left[\frac{a}{k} \right]^2 \right]$

As in the proof of Theorem 5 we get that the average

(8) Note that we are being rather loose here. These are the equilibrium values of Θ when q is large; the equations to follow hold asymptotically, with probability 1.

flow-time of these passengers is

$$F_q = \frac{b^2 p q}{k} \left[\sqrt{a_d} + \sqrt{\frac{a_s}{k}}^2 (1+1/2m) \right]$$
 (4.8)

for q large. This approximation to the value of F_q should not be taken too literally: some of its limitations are discussed in Section 4.3.

Various other similar schemes might also be suggested for serving the (R_s,R_d) and (R_d,R_s) demands. Suppose that the number of buses is increased. Then, the average flow-time will be minimized if we increase also the number of regions of R_s , and retain a single bus in each. However, it might also be desirable to have more than one bus in each region r_i . For example, two buses travelling through the subregions as shown in Figure 4.7 would provide a more equitable service to passengers (i.e. s or variance of travel-time) and a lower value of the ma hum flow-time. This scheme is also more reliable: if a bus breaks down, its passengers can still be collected by the other bus. These are all considerations that are important in practice.

Having described how passengers travelling between R_d and R_g might be served, we now turn to suburban demands, of the form (R_g, R_g) . In general we have seen that a service with transfers will yield an average flow-time that is better than that with a transfer-free service. So, consider a scheme with transfers.

-96-



Figure 4.7 Route taken by a pair of buses in a region

With j buses, partition R_s into j regions $r_1, \ldots r_j$ as was done before, and assign a bus to each. Also specify a transfer point in each region and a route for a line-haul bus, travelling between these transfer points, in the manner of the 'Michigan Scheme', as described in Section 1.1. This line-haul bus will play the role of a transfer point: the distance which it travels is asymptotically negligible, and it is clearly not always practical to have all regional buses meeting at the same transfer point as was required in Algorithm 3. For our example, this is sketched in Figure 4.8.

Once more, subdivide the regions r_1 ; the regional buses perform simple tours in their subregions in the same manner as those before, visiting the transfer point each period. The previous remarks on the number of buses in each

-97-





region are again pertinent.

If there are j regional buses and j regions, with m subregions each, the asymptotic average flow-time can be calculated from Theorem 6 (set k=j; h=1) to be

$$\mathbf{F}_{\mathbf{q}} = \frac{\mathbf{b}^{2}(1-\mathbf{p})\mathbf{q}\mathbf{a}_{\mathbf{B}}}{\mathbf{j}^{2}} \begin{bmatrix} 3+1-1(1-1/\mathbf{m})\\ \mathbf{m} & \mathbf{j} \end{bmatrix} + \mathbf{Q}_{1} \qquad (4.9)$$

Here, Q_1 is the average time spent by a passenger on a fixed-route bus - it depends for example upon the number of these buses and upon the actual route taken.

This scheme corresponds to a single transfer per period. It is possible to obtain further improvement in average flow-time by having, say, two transfers per period. Then, we need 2j regions of R_g , with each bus assigned to two of them, and a transfer point in each. A bus visits its two regions alternatingly, performing a tour through the subregions in each, and visiting the transfer point

-98-

thereafter (in the manner of the tour $T_{2,m}^{j}$ of Algorithm 6). The asymptotic average flow-time can be calculated as before to be

$$F_{q} = \frac{b^{2}(1-p)qa_{s}}{j^{2}} \left[2.5 + \frac{1}{2m} - \frac{1}{2j}(1-1/m) \right] + Q_{2} \quad (4.9a)$$

With this approach, then, it is possible to calculate the asymptotic flow-time and other performance criteria for various suggested schemes as in (4.8) and (4.9). It is indicated in Section 4.3 how adjustments in these formulas might be made to reduce their error. Then, a parametric study can be made to analyse the behaviour of alternative system designs. For example, given a fixed number of buses, determine how they should be allocated between the suburban and central subsystems. Or, calculate the possible gain in flow-time if the number of buses is increased or if there are more transfers on the suburban subsystem. Different schemes with alternative decomposition of the area might also be studied, and other performance criteria such as the average waiting-time or maximum travel-time might be investigated. This can all be done analytically, avoiding the need for complex simulations.

CHAPTER V

CONCLUSION

Our first aim in this report has been towards the development of dial-a-ride scheduling algorithms. We have attempted an analysis of the problem at its most fundamental level, i.e. to question what design approach should be taken. In our study we have viewed the system macroscopically - by considering the statistics of the arriving demands, we have analysed probabilistically the combinatorial aspects of the problem for suitable mathematical models. This has led to a derivation of a class of algorithms (or, schemes) for which performance can easily be evaluated, and hence to an approach towards the design of dial-a-ride systems.

The proposed approach, then, is justifiable in a mathematical sense. The justification is with respect to simple criteria that are pertinent both to the operator of the system and to the passengers. The mathematical sense is asymptotic and probabilistic, and will be discussed shortly.

From a practical viewpoint, the design approach has many attractive features. Most important is that the schemes have a decomposed, modular nature. This yields a remarkable simplicity at many levels. Computational

DEDING PAGE BLANK-NOT FILMED

requirements are kept low: no long searches are needed, and tours for the buses are calculated at discrete instants of time in a decentralized fashion. The systems are simple and easy to visualize: passengers should understand their basic operation and anticipate the type of service to be provided. It is possible too to provide incentives for the bus-drivers: if each driver performs as well as he is able, good performance for the whole system will result.

A further feature is that a tool for design has been provided. It is possible to calculate analytically the dynamic behaviour of each bus, and hence to measure the performance of a suggested scheme in terms of the parameters (e.g. the number of buses, the size of the region, the number of transfer points, and so on). One can thus choose parametric values 'optimally' and can evaluate and compare alternative schemes.

As remarked, we have concerned ourselves with the fundamentals of algorithmic design. Therefore, we have had to avoid operational details - there are very many practical aspects which we have barely, if at all, touched upon. So, there is scope for much additional work. Further analytic work could focus upon a specific methodology for choosing a decomposition of the region and a specialization of the buses in subregions, particularly when the demand is nonuniform. We have but indicated (in Chapter IV) how the design tool might be sharpened and calibrated if stochastic disturbances, low demand rates and additional travel-times are considered. A computer simulation might be used to develop these methods. A final validation of the approach would be in a real implementation: the schemes do appear promising since a related system at Ann Arbor, Michigan, has been successful.

The approach has been justified via the mathematical models of Chapters II and III. This brings us to the second concern of our report: Scheduling Theory as studied in Operatic Research.

he start, we wished to question the asymptotic approach to combinatorial optimization suggested by Karp [9]. Karp's approach has permitted a global view of the problem: when a new demand arises there is no need for a large combinatorial manipulation of the existing solution, and a small adjustment suffices. It yields simple, decentralized algorithms that are efficient (in terms of computational requirements) and yet are nontrivial. Optimality can be essentially guaranteed when the problem size is very large. When the problem size is small, the theory is no longer valid, but the resulting algorithms are nonetheless relevant. Thus, we are led to claim that Karp's approach does appear to have practical worth.

In applying [9] to our transportation problem we have generalized a result of Beardwood et. al. [2]. It is tempting to conjecture that similar generalizations, yielding interesting practical algorithms, might be possible for other travelling salesman-like problems that abound in Operations Research.

Traditionally, scheduling has been regarded as a combinatorial problem. This emphasis has produced largely intractable problem formulations and, except for some special cases (see Conway et. al. [6]), few insights. Our study has essentially avoided combinatorics. It is believed that many of the features we have obtained are more generally relevant to scheduling applications. Particularly when there are stochastic disturbances (as is the case in most large real problems) it is felt that combinatorial aspects can be dissipated in a fairly simple way, by decomposing and discretizing the problem suitably. Then, effects predominate: these are problems of other coordination and of designing the schedule in advance. The schedule should be 'stable' with respect to reasonable disturbances and should be 'flexible' so that changes can easily be made when required. Phenomena such as these are not well understood within the present framework of Scheduling Theory.
APPENDIX TO CHAPTER II

This appendix is a collection of detailed proofs of the assertions in Chapter II. All notation and terminology is as described there. For the most part the results are proved rigorously; for the sake of clarity in Proposition 3.2 we do revert to a more informal presentation.

First we make some preliminary notational remarks, and restate the law of large numbers and Beardwood's result for completeness.

Preliminary Remarks

Throughout the appendix we shall be dealing with random variables. We must explicitly identify the underlying probability space and describe our notation.

Let $w = \{(o_1, d_1), (o_2, d_2), \ldots\}$ be a countable sequence of pairs of points in the planar region R. Let $w_n = \{(o_1, d_1), \ldots, (o_n, d_n)\}$ be the first n pairs of w. The set W of all sequences w is the sample space. If we regard w as a sequence of 'random' pairs, each element of $\{o_1, d_1, o_2, d_2, o_3, d_3, \ldots\}$ being independently and uniformly drawn from the points in R, then we have a probability space. Random variables are all defined on this space. The set of 'problem instances' of size n is {wn, weW}.

For example, $Y_n(w)$ is the length of the optimal bus tour through the first n points of weW, and $Y_n(\cdot)$ is a random variable. If $X_n(\cdot)$, n=1,2,... are random variables then limsup $X_n(\cdot)$ and liminf $X_n(\cdot)$ are well defined $n \rightarrow \infty$ random variables. If X and Y are random variables, $X \leq Y$ means $X(w) \leq Y(w)$ for all weW; $X \leq Y$ almost everywhere (a.e.) means $Pr[X(w) \leq Y(w)] = 1$.

For clarity, in what follows, we shall omit explicit mention of the sample space and the parameterization on the w's.

Kolmogorov's Strong Law of Large Numbers [14]

Let $\{X_1, i=1,2,...\}$ be a sequence of independent identically distributed random variables, such that $E(X_1) < \infty$. Then

$$\frac{1}{\Sigma} \sum_{i=1}^{n} x_{i}$$
 converges almost everywhere n i=1

and

 $\lim_{n \to \infty} \frac{1}{\Sigma} \sum_{i=1}^{n} X_{i} = E(X_{1}) \qquad \text{a.e.}$

2.1 The Travelling Salesman Problem

Theorem 1 (Beardwood et. al. [2])

Let R be a planar Lebesgue set with area a>0 and let L_n be the length of the shortest path through n points which are uniformly and independently distributed in R. There exists an absolute constant b, independent of the problem instance and of the shape of R such that

 $\lim_{n \to \infty} \frac{L_n}{\sqrt{n}} = b\sqrt{a} \qquad \text{almost everywhere (a.e.)}$

Let us prove an easy initial lemma that will be required in many proofs to follow: if the number of points in a certain subregion is random, we can replace this random number by its asymptotic value in Beardwood's formula.

Lemma 1.0

Consider a particular region of area a in which there are p_n points; p_n is a random variable satisfying for some constant g,

$$\lim_{n \to \infty} \frac{p_n}{n} = g \quad \text{a.e.} \qquad (A1.0)$$

Let h_n be the length of an optimal travelling salesman tour on these p_n points. Then,

$$\lim_{n \to \infty} \frac{h_n}{\sqrt{n}} = b\sqrt{g}\sqrt{a} \qquad a.e$$

Proof

By Theorem 1, we have

$$\lim_{p_n \to \infty} \frac{h_n}{\sqrt{p_n}} = b\sqrt{a} \quad a.e. \quad (A1.1)$$

-107-

By (A1.0),

$$\lim_{n \to \infty} \frac{p_n}{n} = \sqrt{g} \quad a.e. \qquad (A1.2)$$

[]

Let any $\varepsilon > 0$ be given. Let $\varepsilon_1 = \max\{\varepsilon/3, \varepsilon/3b\sqrt{a}\}$ and $\varepsilon_2 = \max\{\varepsilon/3, \varepsilon/3\sqrt{g}\}$. From (A1.2) we have that there exists N₁ such that for $n \ge N_1$,

 $(\sqrt{g}-\varepsilon_1)\sqrt{n} \leq \sqrt{p_n} \leq (\sqrt{g}+\varepsilon_1)\sqrt{n}$ a.e. From (A1.1), there exists N₂ such that for $n\geq N_2$, $(b\sqrt{a}-\varepsilon_2)\sqrt{p_n} \leq h_n \leq (b\sqrt{a}+\varepsilon_2)\sqrt{p_n}$ a.e.

Thus, for $n \ge \max\{N_1, N_2\}$,

$$(b\sqrt{a}-\epsilon_2)(\sqrt{g}-\epsilon_1) < \frac{h_n}{\sqrt{n}} < (b\sqrt{a}+\epsilon_2)(\sqrt{g}+\epsilon_1) \quad a.e.$$

i.e. $b\sqrt{a}\sqrt{g}-\epsilon < \frac{h_n}{\sqrt{n}} < b\sqrt{a}\sqrt{g}+\epsilon \quad a.e.$

and the result follows.

2.2 Static Single-Bus Problem

Observation 2.0

The single-bus static problem is NP-complete. <u>Proof</u> (See Aho [1] for the notation.)

It is easy to see that the bus problem is in NP.

Given an instance of the TSP of size n, we can transform it in polynomial time into an instance of the bus problem of size n as follows. Let the given n points be origins $\{o_1, \ldots o_n\}$, and let all have as common destination a point p, a distance of at least DA from every origin, where D is a very large real number and A is the $\max\{\|o_i - o_j\|, i, j=1, \dots\}. For D \text{ large enough, an optimal} \\ \text{bus tour will visit all origins first, and only then visit} \\ \text{p. Clearly, the subtour visiting all origins is an optimal} \\ \text{open travelling salesman tour.} \qquad []$

For the proof of Lemmas 2.2 and 2.3 we require an easy preliminary lemma.

Lemma 2.1

$$\frac{2m}{1} \sum_{m} \sqrt{1} = \frac{4}{\sqrt{2}} + 0(1/m)$$

Proof



Figure A2.1 Diagram for Lemma 2.1

i.e.
$$\frac{1}{m^{3/2}} \begin{bmatrix} 2(2m)^{3/2} \\ 3 \end{bmatrix} \leq \frac{1}{m^{3/2}} \frac{\sum_{i=1}^{2m} \sqrt{i}}{\sum_{i=1}^{3/2} \frac{\sum_{i=1}^{2m} \sqrt{i}}{m^{3/2}} \leq \frac{1}{m^{3/2}} \begin{bmatrix} 2(2m+1)^{3/2} - 2 \\ 3 \end{bmatrix}$$

i.e. $\frac{2}{3} \frac{2^{3/2}}{\sum_{i=1}^{3/2} \frac{\sum_{i=1}^{2m} \sqrt{i}}{\sum_{i=1}^{3/2} \frac{\sum_{i=1}^{2m} \sqrt{i}}{3}} \leq \frac{2(2+1/m)^{3/2}}{3m^{3/2}} - \frac{2}{3m^{3/2}}$
i.e. $\frac{4}{3} \sqrt{2} \leq \frac{1}{m^{3/2}} \frac{\sum_{i=1}^{2m} \sqrt{i}}{\sum_{i=1}^{3} \sqrt{i}} \leq \frac{4}{3} \sqrt{2} + 0(1/m)$ []

Lemma 2.2

 $\limsup_{n \to \infty} \frac{Y_n}{\sqrt{n}} \leq \frac{4}{\sqrt{2}} b \sqrt{a} \quad a.e.$

Proof

Consider the feasible bus tour $T_{I\!I\!I}^O$ given by Algorithm 2.

- Let \bar{o}_1 be the number of origins in each region r_1
 - d_i be the number of destinations in r_i that are visited on the first passage (with origins in regions $r_1, r_2, \dots r_{i-1}$)

 \bar{d}_1 be the number of destinations in r_1 that are visited on the second passage through the regions. Let s_1^1 be the length of the shortest tour through the $\bar{o}_1 + \bar{d}_1$ points in r_1 visited on the first passage, and s_1^2 be the length of the shortest tour through the \bar{d}_1 points visited on the second passage - these are random variables.

-110-

If Y_n is the length of the optimal bus tour then

$$Y_n \leq \sum_{i=1}^{m} s_i^1 + \sum_{i=1}^{m} s_i^2 + 2m\Delta(R)$$
 (A2.1)

where $\Delta(R)$ is the diameter of R (so $\Delta(R)/\sqrt{n} \rightarrow 0$ as nto).

Now, the law of large numbers yields, as $n \rightarrow \infty$,

$$\frac{\overline{o_1}}{n} \rightarrow \underbrace{1}_{m} \qquad \text{a.e.}$$

$$\frac{\overline{d_1}}{n} \rightarrow \underbrace{(1-1)}_{m^2} \qquad \text{a.e.}$$

$$\frac{\overline{d_1}}{n} \rightarrow \underbrace{m-i+1}_{m^2} \qquad \text{a.e.}$$

and

Thus, by Theorem 1 and Lemma 1.0,

$$\lim_{n \to \infty} \frac{s_1^1 + s_1^2}{\sqrt{n}} = b \int_{m}^{1 + (i-1)} \int_{m}^{a} + b \int_{m}^{m-i+1} \int_{m}^{a} a.e.$$
$$= \frac{b\sqrt{a}}{m^{3/2}} (\sqrt{m-1+i} + \sqrt{m-i+1}) \quad a.e.$$

Hence.

$$\lim_{n \to \infty} \frac{1}{\sqrt{n}} \begin{bmatrix} \frac{m}{\Sigma} s_1^1 + \frac{m}{\Sigma} s_1^2 \end{bmatrix} = \frac{b\sqrt{n}}{\sqrt{n}} \begin{bmatrix} \frac{2m}{\Sigma} \sqrt{1} + \sqrt{m} - \sqrt{2m} \end{bmatrix} \quad \text{a.e.}$$

Inequality (A2.1) now yields

$$\limsup_{n \to \infty} \frac{Y_n}{\sqrt{n}} \leq \frac{b\sqrt{a}}{m^{3/2}} (\sum_{i=1}^{2m} \sqrt{i}) + \frac{b(1-\sqrt{2})}{m} \sqrt{a}$$

This holds for arbitrary m, so we can let $m \rightarrow \infty$, and we obtain by Lemma 2.1,

$$\limsup_{n \to \infty} \frac{Y_n}{\sqrt{n}} \leq \frac{4}{3} \sqrt{2} b \sqrt{a} \qquad a.e.$$

1

Lemma 2.3

 $\liminf_{n \to \infty} \frac{Y_n}{\sqrt{n}} \geq \frac{4}{\sqrt{2}} \sqrt{2} \sqrt{a} \qquad a.e.$

Proof

First, some notation. In any problem instance, let O_n be the set of n origins and D_n be the set of n destinations. Let T_n be any optimal tour (which has length Y_n). By this we mean that T_n is that connected subset of R which is traversed in visiting $O_n \cup D_n$. A tour segment through P (where $P \subseteq T_n$) is the minimal connected subset of T_n that contains P.

The optimal tour T_n induces a feasible ordering on the points of $O_n \cup D_n$. Let us label the origins in O_n such that

 $i \langle j \rangle \langle z \rangle \circ_i$ proceeds \circ_j on the tour T_n . Let d_i be the destination corresponding to \circ_i .

Let m be any integer, m>1. We partition R into m Lebesgue subregions, each of area a/m.

Let $(T_n)^1$ be the tour segment through $\{o_1, o_2, \ldots o_{\lfloor n/m \rfloor}\}$, where $\lfloor n/m \rfloor$ is the smallest integer larger than n/m. (The segment $(T_n)^1$ possibly visits some destinations en-route.) Similarly, let $(T_n)^j$ be the tour segment through $\{o_{\lfloor (j-1)n/m \rfloor+1}, \ldots o_{\lfloor jn/m \rfloor}\}$. There exist partitions of R into m Lebesgue subregions $\hat{R}_1, \ldots \hat{R}_m$ each with area a/m and bounded perimeter (length of boundary) such that the origins in \hat{R}_j are precisely those in $(T_n)^j$, for $j=1,\ldots m$. Let $A_1,\ldots A_m$ form a partition

of R drawn at random from the space of such partitions⁽¹⁾. We can assume that the origins in A_j are uniformly distributed there.

Let us define for each subregion A_j , j=1,2,...m, $\overline{o}_{nj} = number of erigins in <math>A_j = [n/m]$ or [n/m]+1 $\overline{d}_{nj} = number of destinations in <math>A_j$ $\overline{d}'_{nj} = number of destinations in A_j$ which also belong to $\{d_1, d_2, ..., d_{[nj/m]}\}$ - i.e. destinations in A_j whose corresponding origins are in $A_1 \cup A_2 \cup ... \cup A_j$.

(1) An example of suitable A_j , $j=1,\ldots m$, is provided by the following.

Let B_i , i=1,...n, be a ball around o_i with radius $\tilde{\epsilon} < 1/n^2$ and with $B_i \cap B_j = \emptyset$ for $i \neq j$. That is,

 $\tilde{\epsilon} < \min\{1/n^2, \min\{\|o_1-o_j\|/2, i, j=1, ..., i\neq j\}\}$ and $B_1 = R \cap \{x, \|x-o_j\| < \tilde{\epsilon}\}, i=1, ... n.$

Now partition $R-\bigcup_{i=1}^{U} B_i$ randomly into m equal regions i=1 Xj, j=1,...m, such that each Xj is a Lebesgue set with perimeter less than a fixed, suitably large number, K.

Finally, let A_j be X_j together with the balls around the origins in $(T_n)^j$, i.e.

 $A_j = X_j U B_{[(j-1)n/m]+1} U \dots U B_{[jn/m]}$ The perimeter of A_j is less than

 $K + n.2\pi/n^2 = K + 2\pi/n$

(This is a bounded function of n, as is to be required in (A2.6a).)

 \bar{d}_{nj}^{*} = number of destinations in $A_{j} \cap (T_{n})^{j}$ - i.e. destinations which are visited by the segment

 $(T_n)^j$ in Aj. Note that $\bar{d}_{nj}^n \leq \bar{d}_{nj}^n$.

-

Since the regions A_j were randomly selected, and the destinations were uniformly distributed over R, the random variables \bar{d}_{nj} and \bar{d}'_{nj} satisfy, by the law of large numbers,

anj	\rightarrow	1	a.e.	as	n→∞
n		10			
d'nj	→	<u>j</u>	a.e.	as	n→∞
-		n -			
onj	\rightarrow	1	as n→∞		
n		10			

Also,

Thus, given any $\varepsilon > 0$, there exists $N_1 > 0$ such that, for $n \ge N_1$,

 $-\varepsilon < \frac{\bar{d}_{nj}}{n} - \frac{1}{m} < \varepsilon$ a.e. (A2.2)

$$-\varepsilon < \frac{d_{nj}}{n} - \frac{j}{m^2} < \varepsilon$$
 a.e. (A2.3)

and

 $-\varepsilon < \frac{\bar{o}_{nj}}{n} - \frac{1}{m} < \varepsilon$ (A2.4)

Now, the tour within A_j , viz. $T_n \cap A_j$, consists of one or more connected pieces. Discard any such piece which does not contain a point of $O_n \cup D_n$. The closure of any surviving piece will be called a j-piece. The set of j-pieces can be partitioned into two parts (see Figure A2.2):

 $j^{(1)}$ -pieces are those that belong to $(T_n)^j$ $j^{(2)}$ -pieces are the remainder.



Figure A2.2 The Tour $T_n^* - (i)$

Part of the optimal tour T_n^{\bigstar} is shown. The arrows indicate the direction traversed.

 $j^{(1)}$ -pieces are shown ----- $j^{(2)}$ -pieces are shown ++++++ $(T_n)^j$ -segment is the ---- and ==== path.

Thus, the $j^{(1)}$ -pieces contain all \tilde{o}_{nj} origins in Aj as well as the $\bar{d}_{nj}^{"}$ destinations. The $j^{(2)}$ -pieces contain $\bar{d}_{nj} - \bar{d}_{nj}^{"}$ destinations.

Let $s_j^{(1)}$ be the length of the $j^{(1)}$ -pieces $s_j^{(2)}$ be the length of the $j^{(2)}$ -pieces.

Clearly,

 $Y_n \ge \sum_{j=1}^{m} (s_j^{(1)} + s_j^{(2)})$ (A2.5)

Let $s_j^{e(1)}$ be the length of an optimal travelling salesman tour through the $\tilde{o}_{nj} + \tilde{d}_{nj}^{*}$ points of the

j⁽¹⁾-pieces.

Let $s_j^{\pm(2)}$ be the length of an optimal travelling salesman tour through the $\bar{d}_{nj} - \bar{d}_{nj}^{"}$ points of the $j^{(2)}$ -pieces.

For k=1,2, Beardwood et. al. (their Lemma 2) have shown that there exists a $\Delta_j^{(k)}$ such that

$$\frac{\Delta_j^{(k)}}{\sqrt{n}} \rightarrow 0 \quad \text{as } n \rightarrow \infty \qquad (A2.6a)$$

satisfying

$$s_{j}^{*(k)} \leq s_{j}^{(k)} + \Delta_{j}^{(k)}$$
 (A2.6)

This is obtained by constructing a feasible travelling salesman tour through the points of the $j^{(k)}$ -pieces, each traversed just once, together with a part of the double circuit of the boundary of A_j. (An example is diagrammed in Figure A2.3.)

Now, there exists an N₂ such that, for all $n \ge N_2$,

$$0 < \frac{\Delta_j^{(k)}}{\sqrt{n}} < \frac{\varepsilon}{m}$$
 for k=1,2; j=1,2,...m. (A2.7)

By Theorem 1, for n>N2,

$$b \int_{m}^{a} - \frac{\varepsilon}{m} \leq \frac{s_{j}^{*}(1)}{(\bar{o}_{nj} + \bar{d}_{nj}^{*})^{1/2}}$$
 a.e. (A2.8)

and

$$\sum_{m} \frac{1}{m} - \frac{\varepsilon}{m} \leq \frac{s_{j}^{*}(2)}{(\bar{d}_{nj} - \bar{d}_{nj}^{*})^{1/2}} \quad \text{a.e.} \quad (A2.9)$$

From (A2.8), (A2.9), (A2.2) and (A2.4), we get for $n>N = max\{N_1, N_2\}$,



Figure A2.3 The Tour T_n^* -(ii)

 $j^{(2)}$ -pieces are shown +++++ The tour ---- is an upper bound to the length $s_j^{*(2)}$, an optimal travelling salesman tour through the destinations in the $j^{(2)}$ -pieces.

$$s_{j}^{e(1)} + s_{j}^{e(2)} \geq b \left[\overline{a} \left[\overline{o}_{nj} + \overline{d}_{nj}^{*} \right]^{1/2} + b \left[\overline{a} \left[\overline{d}_{nj} - \overline{d}_{nj}^{*} \right] \right]^{1/2} - \frac{\varepsilon}{m} \left[\overline{o}_{nj} + \overline{d}_{nj}^{*} \right]^{1/2} - \frac{\varepsilon}{m} \left[\overline{d}_{nj} - \overline{d}_{nj}^{*} \right]^{1/2} - \frac{\varepsilon}{m} \left[\overline{a}_{nm} - n\varepsilon - \overline{d}_{nj}^{*} \right]^{1/2} - \frac{\varepsilon}{m} \left[\overline{a}_{nm} - n\varepsilon - \overline{d}_{nj}^{*} \right]^{1/2} - \frac{\varepsilon}{m} \left[\frac{n + 2n\varepsilon + nj}{m} \right]^{1/2} - \frac{\varepsilon}{m} \left[\frac{n + \varepsilon n}{m} \right]^{1/2} - \frac{\varepsilon}{m} \left[\frac$$

So, since $\bar{d}_{nj}^{*} \in [0, \bar{d}_{nj}]$, and with (A2.3) we have

-117-

$$\mathbf{s}_{j}^{*(1)} + \mathbf{s}_{j}^{*(2)} \geq \mathbf{b} \begin{bmatrix} \mathbf{n} - \mathbf{n}\varepsilon + \mathbf{d}_{n,j} \end{bmatrix}^{1/2} + \mathbf{b} \begin{bmatrix} \mathbf{n} - \mathbf{n}\varepsilon - \mathbf{d}_{n,j} \end{bmatrix}^{1/2} \\ - \frac{\varepsilon}{\mathbf{m}} \begin{bmatrix} \mathbf{n} + 2\mathbf{n}\varepsilon + \mathbf{n}\cdot\mathbf{j} \\ \mathbf{m} \end{bmatrix}^{1/2} - \frac{\varepsilon}{\mathbf{m}} \begin{bmatrix} \mathbf{n} + \varepsilon\mathbf{n} \\ \mathbf{m} \end{bmatrix}^{1/2} \quad \mathbf{a} \cdot \mathbf{e} \cdot \mathbf{e} \\ (A2.9a) \end{cases}$$

$$\geq b \left[\frac{\mathbf{a}}{\mathbf{m}} \begin{bmatrix} \mathbf{n} - 2\mathbf{n} \,\varepsilon + \mathbf{n} \\ \mathbf{m} \end{bmatrix} \right]^{1/2} + b \left[\frac{\mathbf{a}}{\mathbf{m}} \begin{bmatrix} \mathbf{n} + \mathbf{n} \\ \mathbf{m} \end{bmatrix} \right]^{1/2} - \frac{\varepsilon}{\mathbf{m}} \begin{bmatrix} \mathbf{n} + \varepsilon \mathbf{n} \\ \mathbf{m} \end{bmatrix} \right]^{1/2} = \frac{\varepsilon}{\mathbf{m}} \begin{bmatrix} \mathbf{n} + \varepsilon \mathbf{n} \\ \mathbf{m} \end{bmatrix}^{1/2} = \mathbf{a} \cdot \mathbf{e} \cdot$$

From (A2.5),

$$\frac{\chi_n}{\sqrt{n}} \geq b \left[\frac{a}{m} \frac{1}{m^2} \sum_{j=1}^{m} (\sqrt{m+j} + \sqrt{m-j}) - \frac{c}{m} \sum_{j=1}^{m} \left[\frac{1+2c+j}{m} \frac{1}{m^2} \right]^{1/2} - c \left[\frac{1+c}{m} \right]^{1/2} \quad \text{a.e.}$$

But, ε is arbitrary, and the coefficients of ε are bounded above, so

$$\liminf_{n \to \infty} \frac{Y_n}{\sqrt{n}} \ge \frac{b\sqrt{a}}{m^{3/2}} \sum_{j=1}^m (\sqrt{m+j} + \sqrt{m-j}) \quad a.e.$$

$$= b\sqrt{a} \begin{bmatrix} \frac{1}{\sum} \sqrt{j} - 1 \\ \frac{1}{m^{3/2}} \end{bmatrix}$$

This holds for any m. Thus, by Lemma 2.1,

$$\begin{array}{cccc} \liminf \underline{Y}_n & \geq & \underline{4}\sqrt{2}b\sqrt{a} & a.e. \\ n \rightarrow \infty & \sqrt{n} & 3 & & \end{array}$$

2.3 Static Multiple-Bus Problems Lemma 3.1 $\lim_{n \to \infty} \frac{Z_n^k}{\sqrt{n}} = c\sqrt{a} \quad a.e.$ Proof Define the random variables \overline{c}^k and \underline{c}^k as follows. $\bar{c}^k = \limsup_{n \to \infty} \frac{Z_n^k}{\sqrt{n}}$ $c^{k} = \liminf \frac{z_{n}^{k}}{\sqrt{n}}$ Note first that it is feasible to have k-1 buses idle and to use only one. Thus, $z_n^k \leq Y_n$ (Yn is as in Theorem 2.) Hence, $\bar{c}^k \leq c$ a.e. by Theorem 2. We now show that $c \leq \underline{c}^k$ a.e. With the fact that $\underline{c}^k \leq \overline{c}^k$, we shall have the required result. Define for i=0,1,2,... auxilliary problems as follows. <u>'Problem i'</u>: Let $Y_n^k(i)$ be the minimal total distance travelled by k buses through n random demand pairs, allowing at most i transfers of passengers - $Y_n^k(i)$ is a random variable. $\limsup_{n \to \infty} \frac{Y_n^k(1)}{\sqrt{n}} = \bar{c}^k(1)$ Let $\underset{n \to \infty}{\text{liminf}} \frac{Y_n^k(i)}{\sqrt{n}} = \underline{c}^k(i)$

Now, for 1=0,1,2,..., we have

-119-

 $Y_n^k(i) \geq Y_n^k(i+1)$

and

 $Y_n^k(i) \geq Y_n^k$.

So,

 $\underline{c}^{k}(i) \geq \underline{c}^{k}(i+1)$

and also

 $\underline{c}^{k}(i) \geq \underline{c}^{k}$ for all i.

Thus, the $\{\underline{c}^{k}(i), i=0,1,2,...\}$ form a monotonically decreasing sequence in i, bounded below by \underline{c}^{k} . Thus, $\lim_{k \to \infty} \underline{c}^{k}(i) \stackrel{\Delta}{=} \overline{\Phi}^{k}$ exists, with $\overline{\Phi}^{k} \ge \underline{c}^{k}$.

It is shown in the Lemma 3.1.1 below that

 $\bar{\boldsymbol{\varphi}}^{\mathbf{k}} = \underline{\boldsymbol{\varphi}}^{\mathbf{k}} \tag{A2.10}$

For any fixed i, consider the optimal tour-length $Y_n^k(i)$ - a random variable. It is possible to justify - see Lemma 3.1.2 below - that

$$Y_n^k(i) + (i+1)(k-1)\Delta \ge Y_n^{\dagger}$$
 (A2.11)
where Y_n^1 is the optimal distance travelled by a single

bus, and Δ is the diameter of the region R.

Dividing (A2.11) by \sqrt{n} and letting $n \rightarrow \infty$, we obtain by Theorem 2 that

 $\underline{c}^{k}(i) \geq c$ a.e. This holds for any i; thus,

∯^k ≥ c a.e.

and, by (A2.10),

c^k ≥ c a.e.

This proves the result, modulo the two lemmas.

Lemma 3.1.1

We need to show that $\overline{\Phi}^k = \underline{c}^k$. Suppose to the contrary that $\overline{\Phi}^k > \underline{c}^k$ at some wew. Let $\varepsilon = \overline{\Phi}^k - \underline{c}^k > 0$. Now, there exists an I such that for $i \ge I$,

$$\Phi^{\mathbf{k}} - \varepsilon/6 < \underline{c}^{\mathbf{k}}(\mathbf{i})$$

Also, since

$$\frac{c^{k}(i)}{n \rightarrow \infty} = \liminf_{n \rightarrow \infty} \frac{Y_{n}^{k}(i)}{\sqrt{n}},$$

there exists N such that for all $n \ge N$,

 $\frac{\underline{Y}_{n}^{k}(\underline{i})}{\sqrt{n}} \geq \underline{c}^{k}(\underline{i}) - \frac{\varepsilon}{6}$

- see Royden [20, page 37].

Thus, for $i \ge I$ and $n \ge N$,

$$\frac{\underline{Y}_{n}^{k}(\underline{1})}{\sqrt{n}} \geq \underline{\Phi}^{k} - \frac{\varepsilon}{3}$$
$$= \underline{c}^{k} + \underline{2}\varepsilon$$

Since

$$\liminf_{n \to \infty} \frac{Z_n^k}{\sqrt{n}} = c^k,$$

there exists an h>N such that

$$\frac{z_h^k}{\sqrt{h}} < \underline{c}^k + \frac{\varepsilon}{3}$$
 (A2.13)

- again, see Royden [20, page 37].

But Zh is the length of an optimal tour on h demands. This tour can have only a finite number of transfers, so there exists an integer $p_h(\varepsilon)$ such that for almost every problem instance, Z_h^k requires at most $p_h(\varepsilon)$ transfers. Thus, for $i>p_h(\epsilon)$,

 $Z_{h}^{k} = Y_{h}^{k}(i)$ (A2.14)

(A2.12)

Hence, for $n \ge N$ and $i \ge max\{I, p_h(\epsilon)\}$, we have from (A2.12) - (A2.14),

$$\underline{c}^{k} + \underline{2} \epsilon < \underline{c}^{k} + \underline{\epsilon}$$

This is clearly absurd and the lemma is proved. Lemma 3.1.2

We assume that the buses start their tours from certain fixed (but arbitrary) points in R, and that all buses terminate their tours at a common final point in R. Consider the following acyclic directed graph, G, representing the tour with length $Y_n^k(i)$ for any problem instance.

Represent by nodes the k starting points, the transfer points (there are at most i of them) and the final terminal point. The tour of a particular bus i is represented by a path through G: edges into a node represent the buses which transfer passengers there, and the indegree equals the outdegree at all nodes representing transfers. An edge thus represents the path through R taken by the corresponding bus, and G represents the given set of tours with total length $Y_n^k(i)$.

An example best illustrates this definition of G. The graph in Figure A2.4 represents a tour with 7 buses and 6 transfer points.

Suppose the k tours are executed by a single bus in the following way. Pass along the same routes, 'backtracking in a straight line from each node to a previous one as far as is necessary, so that all incoming

-122-



-123-



edges have been executed'. (We do not feel that it is necessary to be more specific here. The idea is precisely the reverse of the standard depth-first-search algorithm see e.g. Abo et. al. [1].) At worst we will have to backtrack k-1 times at each of i+1 nodes (the transfer points and the terminal point), and each backtrack will involve a distance of at most Δ , the diameter of R.

This leads to a feasible single-bus tour with length

 $s \leq Y_n^k(i) + (k-1)(i+1)\Delta$

yielding inequality (A2.11).

Lemma 3.1 is thus proven.

[]

Theorem 3

If transfers are allowed $\lim_{n \to \infty} \frac{Y_n^k}{\sqrt{n}} = \frac{c\sqrt{a}}{k} \quad a.e.$

Proof

 $\sum_{n=1}^{k} x_{n}^{i}$ is the total distance travelled by the optimal tour.

If Z_n^k is as in Lemma 3.1, then

 $\sum_{i=1}^{k} x_{n}^{i} \geq z_{n}^{k}$

Now, $kY_n^k \ge \sum_{i=1}^k x_n^i$ by definition of Y_n^k .

Dividing by \sqrt{n} and letting $n \rightarrow \infty$, we obtain, by Lemma 3.1,

$$\liminf_{n \to \infty} \frac{kY_n^k}{\sqrt{n}} \ge c\sqrt{a} \qquad a.e. \qquad (A2.15)$$

Consider the tour $T_m^{k^O}$ given by Algorithm 3. For this tour let

 $s_n^i(m)$ be the total distance travelled by bus i $h_n^1(m)$ be the total length of the travelling salesman tours covered by bus i in its regions.

Then,

 $s_{n}^{i}(m) \leq h_{n}^{i}(m) + m\Delta$

where Δ is the diameter of the region R.

Now, by Lemmas 1.0 and 2.2,

$$\limsup_{n \to \infty} \frac{h_n^1(m)}{\sqrt{n}} \leq mc \boxed{1a} + O(1/m) \quad a.e.$$

Hence,

$$\limsup_{n \to \infty} \frac{s_n^1(m)}{\sqrt{n}} \leq c\sqrt{a} + O(1/m) \quad a.e.$$

-124-

by the optimality of
$$T_n$$
 we have
 $Y_n^k \leq \max\{s_n^i(m), i=1,2,...k\}$ for any m.
Thus,
limsup $\frac{Y_n^k}{\sqrt{n}} \leq c\sqrt{a} + O(1/m)$ a.e. for any m.
 $n \rightarrow \infty \quad \sqrt{n}$
Letting $m \rightarrow \infty$,
 $\lim_{n \rightarrow \infty} \frac{Y_n^k}{\sqrt{n}} \leq c\sqrt{a}$ a.e. (A2.16)
The inequalities (A2.15) and (A2.16) prove the lemma.
[]

Proposition 3.2

The 3-bus fixed-route scheme, described in Section 2.3.3 yields a time-to-delivery Y_n^{3F} satisfying

 $\lim_{n \to \infty} \frac{Y_n^{3F}}{\sqrt{n}} = \frac{1}{3\sqrt{3}} (\sqrt{3} + \sqrt{2} + 1)b\sqrt{a} \quad \text{a.e.}$ = .798b/a a.e.

Proof

We use the same notation as that used for describing the algorithm in the text. Our derivation will be informal, but hopefully transparent. We need only consider bus I since the other buses have analogous tours and so distances travelled will be the same (asymptotically a.e.).

Bus I first visits regions (r1,r2,...rm). In each ri

[(m-i+1)+m]n/9m² origins are collected and

 $[1-1]n/9m^2$ destinations are delivered.

So, by Lemma 1.0, there is a tour-length of $b\sqrt{a/3m}\sqrt{2n/9m}$ in r1.

... k

Second, the bus visits $(r_{2m}, r_{2m-1}, \dots, r_{m+1})$. In each r_1 $[m+(i-1)]n/9m^2$ origins are collected and

 $[(m-i)+m]n/9m^2$ destinations are delivered.

The tour-length in each of these m regions is $b\sqrt{a/3m} (3m-1)/9m^2 \sqrt{n}$.

Finally, the bus visits $(r_1, \ldots r_m)$ again. In each r_i $(m+1)n/9m^2$ destinations are delivered.

So in r_i there is a tour-length of $b\sqrt{a/3m}\sqrt{(m+1)/9m^2}\sqrt{n}$. The total distance travelled by the bus is thus

 $\frac{b\sqrt{a}\sqrt{n}}{3\sqrt{3}}(\sqrt{2} + \sqrt{3}\sqrt{1} + \sqrt{1+1/m})$ $= \frac{b\sqrt{a}\sqrt{n}}{3\sqrt{3}}(\sqrt{2} + \sqrt{3} + 1) + 0(1/m)\sqrt{n}$ $= \frac{3\sqrt{3}}{3\sqrt{3}}$

Letting $m \rightarrow \infty$ we obtain the required result.

[]

APPENDIX TO CHAPTER III

This appendix collects proofs of the assertions in Chapter III. In contrast with Appendix II, proofs are informal. With additional work the results can be made rigorous - for example, Lemma 2.2 can be used as a model for Lemma 4.1 and Lemma 2.3 as a model for Theorem 4. However, it is believed that this would blur their simplicity.

3.1 The Static, Single-Bus Problem

Lemma 4.1

(1)	$\lim_{n \to \infty} \frac{W_n(T_m)}{\sqrt{n}}$		$\frac{2}{15}$ [8/2 - 7] b/a + 0(1/1	n) a.e.
		=	$.5752b\sqrt{a} + O(1/m)$ a.e	
(11)	$\lim_{n\to\infty}\frac{f_n(T_m^o)}{\sqrt{n}}$	•	$\frac{1}{105} [118\sqrt{2} - 47] b\sqrt{a} + 0$)(1/m) a.e.
		=	1.1417b/a + 0(1/m)	a.e.

Proof

Label the m regions r1, r2, ... rm.

By Lemma 1.0, we can assume that, for any i, j=1,...m, there are n/m^2 demands of the form (r_i, r_j) . Let t_i be the time spent in r_i on the first passage t_i' be the time spent in r_i on the second passage $T = \sum_{i=1}^{m} t_i = time-to-completion of first passage.$ For n large enough we have, with probability 1,

$$t_1 = \frac{b\sqrt{n}}{\sqrt{m+1-1}}$$
 (A3.1)

since n/m origins and $(i-1)n/m^2$ destinations are collected. We can assume that the expected time at which any of these points is visited is $(t_1+t_2+\ldots+t_{i-1}+t_i/2)$.

Similarly,

t

$$i = \frac{b\sqrt{n}}{\sqrt{m-i+1}}$$
(A3.2)

since $(m-i+1)n/m^2$ destinations are visited in each region r_1 . The expected time at which these points are visited is

 $T + t_1' + t_2' + ... + t_1'/2.$

First, let us consider the waiting times. The sum of the waiting times in region i is

$$\frac{i-1}{n} [\Sigma t_j + t_i/2].$$

The total average waiting time is then

$$w_n(T_m^o) = \underline{1} \underline{n} \sum_{j=1}^{m} [\Sigma_{t_j} + t_1/2]$$

n m i=1 j=1

$$= \frac{b\sqrt{n}}{m^{7/2}} \left[\sum_{i=1}^{m} (m-i+1/2)\sqrt{m+i-1} \right]$$

by expanding and using (A3.1).

Using a technique similar to that of Lemma 2.1, it can be shown that this series converges and

$$\frac{1}{2} \sum_{m=1}^{\infty} (m-i+1/2)\sqrt{m+i-1} = \frac{2}{15} [8\sqrt{2} - 7] + O(1/m)$$

= .5752 + 0(1/m)

This proves Equation (i). Equation (ii) is somewhat

more cumbersome.

For destinations which are visited on the first passage, the sum of delivery times is

 $G_{1} = \sum_{i=1}^{m} (i-1) \underline{n} (t_{1} + \dots + t_{i-1} + t_{i}/2)$ i=1 _2 $= \frac{bn^{3/2}}{\sqrt{7/2}} \frac{1}{2} \sum_{i=1}^{m} \sqrt{m+i-1} (m^2 - m - i^2 + 2i - 1)$

Again it can be shown that

$$\frac{G_1}{n^{3/2}} = b \left[\frac{2(2^{5/2}-1) - 1(2^{7/2}-1)}{5} \right] + O(1/m)$$

The sum of delivery times for passengers delivered on the second passage is

$$G_{2} = \sum_{i=1}^{m} [T + t_{i}^{i} + ... + t_{i-1}^{i} + t_{i}^{i}/2] \frac{(m-i+1)n}{m^{2}}$$
$$= \frac{bn^{3/2}}{\sqrt{m^{2}}} [\sum_{i=1}^{m} \sqrt{m+i-1} \frac{m}{m}(m+1) + \sum_{i=1}^{m} (m-i+1)^{5/2}]$$
$$= \frac{bn^{3/2}}{\sqrt{m^{2}}} [\sum_{i=1}^{m} \sqrt{m+i-1} \frac{m}{2} (m-i+1)^{5/2}]$$

Once again it can be shown that

$$\frac{G_2}{3^{1/2}} = b \left[\frac{1}{3} (2^{3/2} - 1) + \frac{1}{7} \right] + 0(1/2)$$

Finally, the average flow time is $(G_1+G_2)/n$, i.e.

$$\frac{f(T_{m}^{o})}{\sqrt{n}} = b \left[\frac{2(2^{5/2}-1) - \frac{1}{7}(2^{7/2}-1) + \frac{1}{3}(2^{3/2}-1) + \frac{1}{7} \right] + O(1/m)$$

$$= \frac{b}{105} \left[118\sqrt{2} - 47 \right] + O(1/m)$$

$$= 1.1417b + O(1/m). \qquad []$$

The proof of Lemma 4.2 follows easily from Lemma 4.1 and the discussion in the text of Section 3.1.

[]

-130-

Theorem 4

Let $f_n^* = \inf\{f_n(T), Tes\}.$

Given any $\varepsilon > 0$, there exists an $M(\varepsilon)$ such that for $m \ge M$ and for n large enough, T_m^0 is ε -optimal for problem (3.1) a.e.

Hence, $\lim_{n \to \infty} \frac{f_n^*}{\sqrt{n}} = 1.1417 b \sqrt{a} \qquad a.e.$

Proof

Throughout this proof we assume that n is very large, so results obtained are asymptotic in n. For simplicity, we drop the parameterization on n.

For any problem instance, let the tour T_m^* minimize $\{f(T), Tes_m\}$. Thus, for all m,

 $f(T_m^*) \leq f(T_m^o).$

Let $\varepsilon > 0$ be given, and suppose that the hypothesis is not true. Then, for a subset W' of W with non-null measure, there exists M, such that for all $m \ge M_1$,

 $f(T_m^0) - f(T_m^0) > 3\varepsilon/2$ (A3.3)

By Lemma 4.3 there exists M_2 such that for all $m \ge M_2$,

 $w(T_m^0) + f(T_m^0) - w^+ + f^+ \leq \epsilon/2 \quad a.e. \quad (A3.4)$ So, for m > max{M₁,M₂},

 $f(T_m^{\oplus}) + w(T_m^{\oplus}) \geq f^+ + w^+$ a.e.

 $\geq w(T_m^0) + f(T_m^0) - \epsilon/2 \text{ a.e. by (A3.4)}$ $\geq w(T_m^0) + f(T_m^0) + \epsilon \text{ a.e. by (A3.3)}$ i.e. $w(T_m^0) - w(T_m^0) \geq \epsilon$ a.e. in W' (A3.5) For any partition of R into m equal subregions we

can define a (nonfeasible) tour S_m^O that is similar to

 T_m^o :- visit the regions successively, as for tour T_m^o , but in addition visit the destinations of the form (r_1, r_1) in each region r_1 on the first passage through the m regions. Clearly,

 $w(S_m^o) > w(T_m^o).$

But, by choosing m large enough - i.e. $m>M_3$, say - we have $\binom{1}{3}$

$$w(S_m^0) - w(T_m^0) < \varepsilon/2 \qquad (A3.6)$$

Let $M = \max\{M_1, M_2, M_3\}$. For any $m \ge M$ consider the tour T_m^{\oplus} for any problem instance. We construct another partition of R. (This is again an extension of the method in Theorem 2; see that proof for notation.)

Divide the tour T_m^* into two segments: S' is the segment from the start of the tour to the delivery of the last origin; S" is the remaining segment containing only destinations. Now divide S' into m subsegments: S' is the first segment from the first origin to the (n/m)th origin; S' is the (i)th segment from the ((i-1)n/m)th origin to the (in/m)th origin. There exists a partition of R into m subregions $ir_1, \ldots r_m$ } such that

 $r_i^{\circ} \cap S_i^{\prime} \neq \emptyset$ (r_i° is the interior of r_i) $r_i^{\circ} \cap S_j^{\prime} = \emptyset$ for $i \neq j$ (Note that S' might cross itself, in which case different segments have points in common.)

(1) Observe that this is precisely the technique used in Theorem 2, Equation (A2.9a). The relationship (A3.6) can be justified in the same way; the formulas are particularly cumbersome, and would obscure the simplicity of the argument. The space of all such partitions is nonempty. Let us draw a partition uniformly from this space, and consider the tours T_m^O and S_m^O respectively on this partition of R. We can assume that the origins and destinations are uniformly distributed within each subregion.

By construction,

$$w(T_m^{\bullet}) \leq w(S_m^{\circ}).$$

So, by (A3.5) and (A3.6),

```
w(T_{m}^{O}) + \varepsilon/2 > w(S_{m}^{O})
\geq w(T_{m}^{\bullet})
> w(T_{m}^{O}) + \varepsilon
```

which holds a.e., in W'. This implies that 1/2 > 1, a contradiction.

Thus, the first part of the hypothesis is true. By letting $m \rightarrow \infty$ we obtain, from Lemma 4.2,

 $\lim_{n \to \infty} \frac{f^*}{\sqrt{n}} = 1.1417 \text{ b}\sqrt{a} \quad \text{a.e.} \qquad []$

3.2 The Dynamic Single-Bus Problem

Lemma 5.1

If the system is in steady-state, then with probability 1, the tour T_m^0 given by Algorithm 5 minimizes, among tours in S_m, the average flow-time of all passengers.

Proof

Label the demand pairs as (o_1,d_1) , i=1,2,...n,... Let t,(i) be the time of arrival of (o_1,d_1) ; let t (o_1) be the time at which o_i is visited; and let $t(d_i)$ be the time at which d_i is visited. The average flow-time of the demands is then

$$\lim_{n \to \infty} 1 \sum_{i=1}^{n} [t(d_i) - t_1(i)]$$
 (A3.7)

Now regard the system as a queuing system (see e.g. [11]). It is convenient to consider two arrival streams - the origins $\{o_i\}$ and the destinations $\{d_i\}$. The arrival epoch of o_i is $t_1(i)$ and its departure epoch is $t(o_i)$. The arrival epoch of d_i is $t(o_i)$ and its departure epoch is $t(d_i)$. Thus, at any time t the 'units' to be 'served' which are present in the system are the points which can feasibly be visited at t.

The average flow-time of all units is

- $F = \lim_{n \to \infty} \frac{1}{2n} \{ \sum_{i=1}^{n} [t(o_i) t_1(i)] + \sum_{i=1}^{n} [t(d_i) t(o_i)] \}$
 - $= \lim_{n \to \infty} \frac{1}{2n} \sum_{i=1}^{n} [t(d_i) t_1(i)]$ (A3.8)

which is half the criterion we wish to minimize in (A3.7). So, minimizing (A3.8) will also minimize (A3.7).

We seek a stationary decision rule, X:- whenever a bus exits from a region, this rule will determine which region is to be visited next and which points are to be visited there. Any stationary such rule will induce a stationary arrival pattern on the destinations.

Given any rule X, let

Ny = average number of units in the system

 T_y = average time between arrivals

 F_v = average flow-time of units.

If the system is in steady-state, a well-known result of Little [13] gives that

 $F_{\chi} = T_{\chi} \cdot N_{\chi}$ a.e. (A3.9) Note that since X is stationary, the average time between arrivals of units is constant, i.e.

 $T_{\chi} = 1/2q$ (A3.10)

(q is the constant arrival-rate of demands).

Further, if n units are visited in a region, then the average time spent on each unit is h_n/n , where h_n is the length of the path visiting these n points. For n large enough, h_n/n will be a decreasing function of n with probability 1. Now, to minimize N_{χ} , the average number of units in the system, visit those which will take the shortest time - i.e. minimize h_n/n . Thus, choose the region with most points and, using an optimal travelling salesman tour, visit all points there. By (A3.9) and (A3.10) this decision rule will also minimize F_{χ} (a.e.).

With (A3.8) the result is proven. []

Theorem 5

Let $F_{q,m}$ be the optimal average flow-time for tours in S_m when the arrival rate of passengers is q. Then,

 $\lim_{q \to \infty} \frac{F_{q,m}}{q} = 2b^2 a(1+1/m) \quad a.e.$

Proof

By Lemma 5.1 the tour T_m^0 minimizes the average

flow-time in S_m , so we calculate the average flow-time of this tour.

Let θ be the time spent in each of the m equal areas⁽²⁾. Then, by Equation (3.3),

 $\lim_{q \to \infty} \frac{\theta}{2} = 2b^2 \underline{a} \quad a.e. \qquad (A3.11)$

Consider a time interval [0,0] during which a particular region is visited. During this time q0 origins and q0 destinations are visited; we can take the average time-of-visitation as $t \approx 0/2$. During each of the previous m intervals, of 0 units each, q0/m origins and q0/m destinations arrived at this subregion - here by 'arrived' we refer to the arrival of units or points at the queuing system as described in Lemma 5.1. We can take the average arrival times of these 2q0/m points as $t_1 = -(i-1/2)0$, for i=1,2,...m. Hence, the average flow-time of all points is

 $\frac{1}{2q\theta} \frac{2q\theta}{m} [\theta + 2\theta + ... + m\theta] = \frac{(m+1)\theta}{2}$

By (A3.8) of Lemma 5.1, the average flow-time of passengers is twice this, i.e.

$$F_{\alpha,m} = (m+1)\Theta$$

and

 $\lim_{q \to \infty} \frac{F_{q,m}}{q} = 2b^2 a(1+1/m) \quad \text{a.e. by (A3.11).}$ []

(2) In the text of Section 3.2 we called this quantity θ_q .

3.3 Dynamic Multiple-Bus Problems

Theorem 6

The tour $T_{h;m}^k$ described by Algorithm 6 is optimal (asymptotically a.e.) for the k-bus minimum flow-time problem when tours are in S_{khm} and at most h transfers are allowed per period.

Let $F_{q;h,m}^k$ be the average flow-time resulting from this tour. Then,

 $\lim_{q \to \infty} \frac{F_{q;h,m}^{k}}{q} = \frac{b^{2}a}{k^{2}} \begin{bmatrix} 2 + \frac{1}{mh}(1+1/k) + \frac{1}{(1-1/k)} \\ h \end{bmatrix} \text{ a.e. (A3.12)}$ Proof

We shall not prove the optimality of the tour $T_{h;m}^{k}$. A proof could proceed along the lines of Lemma 5.1, and the same principle would still be valid: each time that a bus exits from a region, visit next the region with most feasible points, and visit all feasible points there. For q large this will yield the tour of Algorithm 6.2. Further, with probability 1, each bus will spend the same time in each region, and there will be no waiting at the transfer point.

Below, we derive (A3.12) in detail.

Each bus visits m regions between any two transfers. Let θ be the time spent by a bus in any subregion. The period is then hm θ . During this time there are qhm θ new demands, and so each bus must serve 2qhm θ/k points. In each region there are 2qhm $\theta/khm = 2q\theta/k$ points visited. Hence,

$$= b \int \frac{2q\theta}{k} \int \frac{a}{khm}$$
$$= \frac{2b^2qa}{k^2hm}$$

i.e.

Each time a region is visited $q\theta/k$ origins and $q\theta/k$ destinations are visited.

Consider a bus B visiting a region that is the j-th after a transfer point. Of the $q\theta/k$ destinations, $q\theta/k^2$ were collected (i.e. their corresponding origins were visited) by bus B itself, and $(k-1)q\theta/k^2$ were collected by the (k-1) other buses. The $q\theta/k^2$ points were visited uniformly during the last hm time intervals, of length θ each. So the waiting-time for these points is

$$\frac{q\theta}{k^{2}hm} (\theta + 2\theta + ... + hm\theta)$$

$$= \frac{q\theta}{2k^{2}} (hm+1)\theta \qquad (A3.14)$$

The $(k-1)q\theta/k^2$ points were collected uniformly during the hm time intervals before the last transfer, (j-1) intervals previously. Thus, the total waiting-time for these points is

$$\frac{(k-1)q\theta}{k^{2}hm} [j\theta + (j+1)\theta + (j+2)\theta + ... + (hm+j-1)\theta]$$

$$= \frac{(k-1)q\theta}{k^{2}} (hm+2j-1)\theta \qquad (A3.15)$$

The $q\theta/k$ origins all arrived during the last hm time intervals. The total waiting-time for these points is

$$\frac{q\theta}{khm}$$
 (0 + 20 + ... + hm0)

 $= \frac{q\theta}{2k} (hm+1)\theta$

(43.16)

(A3.13)

0

The total waiting-time for all points visited during this j-th region of bus B is the sum of (A3.14), (A3.15)and (A3.16). This is

$$\frac{q\theta}{2k^2}$$
. θ [(hm+1)(k+1) + (hm+2j-1)(k-1)]

and $2q\theta/k$ points were visited.

Summing now for j=1,2,...m and dividing by $2q\Theta m/k$ we obtain the average waiting-time of all points, viz.

$$\frac{\theta}{4km} [m(hm+1)(k+1) + \sum_{j=1}^{m} (hm+2j-1)(k-1)]$$

- $= \underline{\Theta m} [khm+k+hm+1 + khm-mh+mk-m]$ 4km
- $= \frac{\Theta m}{4k} \begin{bmatrix} 2kh + (k+1) + (k-1) \\ m \end{bmatrix}$

For the required average flow-time of passengers we must multiply this by 2. Also, using (A3.13) we get

$$F_{q;h,m}^{k} = \frac{1}{2k} \frac{2b^{2}qa}{k^{2}h} \begin{bmatrix} 2kh + (k+1) + (k-1) \\ m \end{bmatrix}$$
$$= \frac{b^{2}qa}{k^{2}} \begin{bmatrix} 2 + \frac{1}{mh}(1+1/k) + \frac{1}{h}(1-1/k) \\ h \end{bmatrix}$$

This gives the required result.

[]

REFERENCES

- Aho, A.V., J.E. Hopcroft and J.D. Ullman, 'The Design and Analysis of Computer Algorithms', Addison-Wesley, Reading, 1974.
- Beardwood, J., J.H. Halton and J.M. Hammersley, The Shortest Path through Many Points, Proc. Cambridge Philosophical Society, <u>55</u>, pp 299-327, 1959.
- 3. Christofides, N., Worst-Case Analysis of a New Heuristic for the Travelling Salesman Problem, Management Sciences Research Report NO. 388, Carnegie-Mellon University, Pittsburgh, Pennsylvania.
- Christofides, N., and S. Eilon, Expected Distances in Distribution Problems, Operational Research Quarterly, 20, pp 437-443, 1969.
- 5. Coffman, E.G., (ed.), 'Computer and Job-Shop Scheduling Theory', Wiley-Interscience, New York, 1976.
- Conway, R.W., W.L. Maxwell and L.W. Miller, 'Theory of Scheduling', Addison-Wesley, Reading, 1967.
- Dreyfus, S.E., 'Dynamic Programming and the Calculus of Variations', Academic Press, New York, 1965.
- Held, M., and R.M. Karp, A Dynamic Programming Approach to Sequencing Problems, J. Soc. Indust. Appl. Math., <u>10</u>, pp 196-210, 1962.
- Karp, R.M., The Probabilistic Analysis of Some Combinatorial Algorithms, Memo. No. ERL-M58, University of California, Berkeley, 1976.
- Kates, J., 'Para Transit Dispatch Systems', J. Kates Associates Inc., Toronto, Ontario, 1975.
- Kleinrock, L., 'Queuing Systems: Vol. 1, Theory', Wiley, New York, 1975.
- Lin, S., and B.W. Kernighan, An Effective Heuristic Algorithm for the Travelling Salesman Problem, Operations Research, <u>21</u>, pp 498-516, 1973.
- Little, J.D.C., A Proof of the Queuing Formula: L= W, Operations Research, 9, pp 383-387, 1961.
- Loeve, M., 'Probability Theory', Van Nostrand, Princeton, 1963.

-139-

- Newell, G.F., Control of Pairing of Vehicles on a Public Transportation Route: Two Vehicles, One Control Point, Transportation Science, <u>8</u>, pp 248-264, 1974.
- Papadimitriou, C.H., The Euclidean Travelling Salesman Problem is NP-Complete, J. Theoretical Computer Science, to appear.
- Potter, B., Ann Arbor's Dispatching System, presented at the Sixth International Conference on Demand-Responsive Transportation Systems and Other Paratransit Services, March 1976.
- Rebibo, K.K., A Computer Controlled Dial-a-Ride System, in AFCET (ed.), Traffic Control and Transportation Systems, North Holland Publishing Co., Amsterdam, 1974.
- Rosenbloom, S., (ed.), 'Paratransit', Special Report 164, Transportation Research Board, Washington D.C., 1976.
- Royden, H.L., 'Real Analysis', Macmillan, New York, 1963.
- 21. Wilson, N.H.M., et. al., Scheduling Algorithms for a Dial-a-Ride System, Massachusetts Institute of Technology, Urban Systems Laboratory, Technical Report USL TR-70-13, March 1971.
- 22. Wilson, N.H.M., et. al., Advanced Dial-a-Ride Algorithms Research Project: Final Report, Massachusetts Institute of Technology, Dept. of Materials Science and Engineering, Technical Report R76-20, March 1976.
B.MC

Anni. Director Electronice and Computer Sciences Office of Director of Debags Restore and Engineering The Pentagen Washington, DC 20315

Office of the Director of Datesor Recearch and Engineering Information Office Library Branch The Postagen Veshington, DC 20101

ODDB and E Advisory Graup an Electras Devices 201 Varias Birect New York, New York 10014

Chief, 8 and D Division (M6) Defente Comunatications Agency Fachington, DC 20501

Director, Mattanal Security Agarery Fort George G. Maada, Maryland 20765

Institute for Defense Analysis Beisece and Technology Division 60 Army Novy Drive Arlangton, Virginis 22202

Dr. Brickley Deless Advanced Research Alter Technical Library 1000 Wilson Beslevard

Commander US Army Balliettes Research Laborenry Ame IMERD BAD Aberdees, MD 21505 Dr. & Reynolds Delonge Advanted Research Proje Alle: Technical Library 1400 Wilson Baulerard Arlington, Virginia 22209

AF/S DPS The Protogen Techington, DC 20150 Commander Picationy Arreads Atta: EMUPA.75-7-8 Derer, NJ 87851

AFBC (LJ/Mr. Sreing B. Mirman) Andrews AFB Variation, DC 10314 US Army Research Office Ann: Dr. Hermann Robil P. G. Bur 12211 Research Triangle Park, MC 21705

Directorate of Electronics and Verprist HQ APRC/DLC Andrews AFB, Maryland 20114

Directorate of Science IIG APRC/DLA Andrews AFB

LTC 7. W. Gregort Ale Parce Office of Scientific Rectored Brilling ATP Transformer, DC 20332

Commander Fraublitet Arcental Alle: Mr. George, C. Walte, Jr. Depoty Director, Pitman Date Lak. Philodolphia, PA 19131

APCBL/LS

Commander US Army Missile Command Ame Chief, Decompat Section Redutuse Arcensi, AL 15807 APCRL/OPL

APCRL/LG

APCAL (CA)

APCRL/LER

Destantes Library (TILD) Rome Air Dreitepress Coster Griffer AFB, New York 1961

Mr. B. B. Potts, Jr. (BCP) Rems Air Breitenman Center Griffine All. Ren Yarb 11841

Mr. Marray Recordings (BCA) Suma Air Breispinger Contar Griffing AFR Rev Tort 1984

AFAL/TE

AFALTER AFA. Ches statt

Chief Belantiet AFAL/CA Vrigts Patteren AFB, Obis 45433

Friday & L. Pasters Badd Bags of Dissistical Rag.

-

Brenter Electronic Variare Laboratory (BCRM) Alle BMSL-VL-MY Viele Back Missis Rauge, Mit 1

Commander US Army Armament Command Atta: UBSAR RD Reck laland, IL 61201

LTC Richard J. Goven Professor Dept. of Electrical Engineering URAF Academy, Calerade Hildes

AUL/LEE-MAS

APETS Technical Library P.O. Bos 6658, MU 5650 Patrick AFB, Florida 13542

ADTC (DLOBL) Eglis AFB, Florids 32542

NG AMD (R DR/COL Gedden) Bronke AFR, Texas 78235

UBAF Europeas Office of Antrop Research Technical Information Office Bus 14, FPO, New York 09510

Dr. Carl E. Baum AFWL (ES) Kirtland AFB, New Mexico #7117

US Army Research Office Ama Mr. Richard G. Clat P.G. Bos 12211 Research Triangle Park, NC 27705

18 Army Research Office Atta: Dr. Januaris R. Bartis P. G. Bon 12211 Research Triangle Park, NC 21704

UB Army Research Office Ame Dr. Harst Withman P. O. Bus 12211 Research Triangle Park, NC 27709

Commendant Of Army Air Optioner School Ame ATRAD 7 COM

AND ALL OF ALL OTHER

Marter Laterstery, SCOM

MAPSAM/RAL Brents AFR. Teme

HODA (DAMA ART A) Washington, DE 20310

Diractor, Div. of Heuroperchistry Walter Reed Army Battrets of Research Washington, DC 30012 Commander UBASATCOM Fort Monmonth, NJ 87793

Commander US Army R and D Group (Par East) APO San Francisco, CA 96363

Commander US Army Commanications Command Alte: Director, Advanced Concepts Office Furt Heacbuck, AZ 85613

Project Manager ARTAIN EAI Building West Long Branch, NJ STIL

Commander US Army White Sands Mireille Range Alla: STEVS-ID-R White Sandr Mireille Range, NM 68092

Director, TRI-TAC Atta: TT-AD (Mrs. Briller) Fort Messmooth, NJ 07703

Commander UB Army Scentity Agency Atta: LAB D-7 Arlington Hall Station Arlington, VA 22212 Commander US Ariny Communication Com Atte: CC-OPS-PD Fort Hunchura, AZ 85613 Commander US Army Material Development and Residence Command Atte Technical Library, Am 30 35 501 Elconderer Arman Alexandrie, VA 32333

COL Robert Noce Senior Dandardisation Representativ 18 Army Standardisation Group, Canada Gamdian Perce Handquarters Ottowa, Gatario, Canada KLA OK2

Constanting The Description Constant Description of the Standard Standard

Project Managar Haval Underwater Sound Conto Ballatic Missile Defense Program Office Technical Library Ann. DAGS BARP Mr. A. Goldi New London, Cont. 06320 1940 Wissen Bird. U. 22000

Dr. Sidney Ross Technical Director SARFA-TD Frankford Arsonal Philadelphia, PA 19137

Department of Naval Respects Electronic and Bolid Boats Briences Program (Code 427) 605 N. Gunny Arlington, Virginia 22217

Office of Noval Research Code 200 Assistant Chief for Technology 800 N. Quincy Arlington, Virginia 22211

Office of Naval Research Information Beloncos Program (Code 437) 809 N. Quincy Arlington, Virginia 12217 Commander Of Army Missile Command Ann: Dight AS Redetman Aresand, AL 15009

Camerander 19 Army Maserials and Mochanico Xesonreit Cantor Atta: Charl, Motorials Sciences Division Waterieum, MA 02172

Director Office of Maral Research Branch Off 514 South Clark Birect Chicage, filiants 69585 Barting Command and Control Buff College His Acquisitions, Lib. (Nr. 1991 Localization, Lib. (Nr.

San Francisco Areo Office Office of Noval Rosearch Tel Marine Streat, Roun 667 San Francisco, C. 10162 Dr. Ress L. Bingfor Army Mansher, TAC/JBEF W Army Electronics Commune Mall L-TL-B Fort Mansher, NJ 97783

Dr. A Louder Chief Relation Office of Heral Research Branch Office 1899 Rast Groon Stream Pacadens, CA 91101

Director Office of Navel Receased Breach Office 716 Breakfury, 5th Finer New York, New York 10001

Mr. L. W Sumary Reval Electronics Systems Com MC No. 1 511 Jafferson (Drie Highersy Arlington, VA 20100

Mr. B. Fretils Mortil Effortrettes Systems Can HC No. 1 1911 Jafferson Dorts Highway Arlington, VA 2010

Mr. N. Butler Hatter Command Nama Electronics Systems Command NG No. 1 2511 Jefferson Davis Highway Ariagino, VA 20540 Cleveland, Chio 44155 Dr. H. J. Meetlar Neval Air Systems Command JP No. 1 1611 Jefferson Davis Highway Arlington, VA 20160

Capt. R. B. Mooks Noval Son Systems Command NC No. 3 2511 Jefferson Davis Highway Arlington, VA 20362

Commander Naval Surface Waapone Caster Alte: Technical Library Silver Spring, Maryland 20010

Officer in Charge Naval Surface Weapone Center Dabigros Laboratory Dabigros, Virginia 12448

Navel Air Development Center Atte: Technical Library Johanvilla Warminster, Penn. 18976

Commander Navel Aviences Facility Indianapolis, Indiana 46241 Atta: D/035 Technical Library

Norol Missile Contor Tectaical Library Code 5632 2 Point Mugu, CA 93042

Naval Weapone Center Affre: Technical Library Code 513 China Labe, CA 11585

Maval Wespuns Center Arta: Code 6010 China Lake, CA 93555

U.S. Naval Oceanngraphic Office Library - Code 1600 Weshington, DC 25373

Commandant, Marine Corps Scientific Advisor (Code AX) Vashington, DC 20140

Dr. Gerner M. R. Wishler Director, Time Service US Naval Observatory Mase Are, at 34th St. N. W. Washington, DC 10346

Nevel Postgraduate School Technical Library Matterey, California 93944

Noral Electronics Laborator Technical Library Res Diego, California 12/52

Herel Electronice Lab Alla: Code 2600 Sas Diego, CA 12112

Hevel Underson Conter Technical Library San Diege, California 92152

Normal Ship Research and Dav Contor David W. Taylor Code 522 1 Botherda, Maryland 2004

Office of Chief of Naval Operations NASCOM/ME Planning Branch NOP-916D, Pentagen Washington, BC 20160

Mr. P. C. Schwash, s.D. T Refinant Astronomics and Spins Administration Tables Administration

Las Alamas Scientific Laborator Atta: Reports Library P. G. Ben 1943 Las Alamas, New Measure 67644

M. Bane Therates Depris Birecter Sectores des Competer Sciences and Technolog Hotional Duress of Stanfords

Director, Office of Pastal Todasilagy (8 and 8) 10 Pastal Service 11111 Partieum prim

Library -8.51 Burnes of Plandards Acquisition Realder, Colorado #0102

MIT Linesta Laboratory Alla: Library A-662 P.O. Ben 73 Louington, Mass. 62173

Dr. Joy Harris Program Director Dovices and Wares Program NGP 1000 G Birect Washington, DC 20550

Dr. Howard W. Eteol Deputy Director Dir. of Materials Research NBF 1809 C Street Washington, DC 20550

Dess Mitchell ngram Director tid State Physics r. of Materials Research Heats Brisses Foundation 06 G Streat ashington, DC 20550

Director Research Laboratory of Electronics Massachusetta Institute of Technology Cambridge, Massachusette d2199

Director Micromove Research Institute Palytochnic Institute ad New York Long Johand Graduate Contor, Route 110 Parmingdule, New York 11735

Anni. Director Microwawa Research Institute Pelytechnic Institute of New York 33 Jay Bereat Breaklyn, New York 11201

Director Microwsve Laboratory Stanford University Stanford, California 46305

Director Electronics Research Laboratory University of California Berbaley, California 60 720

Director Electronics Sciences Laboratory University of Southers Colliferate Las Angeles, Coliferate 90007

Director Electronics Research Cottor The University of Tomos et Austin Engineering-Sciences Bidg. 112 Austin, Tomas 07012

Director of Laboratorios Dir. of Englavoring and Ap Harvard University Pierce Itali Cambridge, Mass. 02130

Director Columbia Radiation Laborator Dapit. of Physics Columbia University 538 West 120th Street New York, New York 10027

Director Coordinated Science Laborate University of Elizate Urbane, Elizate 61001

Director Stanford Electronics Laborate Stanford University Stanford, California, 9104